UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM S-8 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

BATH & BODY WORKS, INC.

(Exact Name of Registrant as Specified in its Charter)

Delaware

(State or Other Jurisdiction of Incorporation or Organization)

31-1029810 (I.R.S. Employer Identification Number)

Three Limited Parkway Columbus, Ohio 43230 Telephone: (614) 415-7000

(Address, Including Zip Code, and Telephone Number, Including Area Code, of Registrant's Principal Executive Offices)

BATH & BODY WORKS, INC. 401(k) SAVINGS AND RETIREMENT PLAN

(Full Title of the Plan)

Michael C. Wu Chief Legal Officer & Secretary Bath & Body Works, Inc. Three Limited Parkway Columbus, Ohio 43230 Telephone: (614) 415-7000 (Telephone number, including area code, of agent for service)

> With a copy to: Jeffrey P. Crandall Davis Polk & Wardwell LLP 450 Lexington Avenue New York, NY 10017 (212) 450-4000

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer \boxtimes

Accelerated filer \Box

Smaller reporting company \Box

Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 7(a)(2)(B) of the Securities Act.

PART I

INFORMATION REQUIRED IN THE SECTION 10(a) PROSPECTUS

The information specified in Item 1 and Item 2 of Part I of this Registration Statement on Form S-8 (this "Registration Statement") is omitted from this filing in accordance with the provisions of Rule 428 under the Securities Act of 1933, as amended (the "Securities Act") and the introductory note to Part I of the Registration Statement on Form S-8. The documents containing the information specified in Part I will be delivered to the participants in the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan (f/k/a the L Brands, Inc. 401(k) Savings and Retirement Plan") covered by this Registration Statement as required by Rule 428(b)(1).

PART II

INFORMATION REQUIRED IN THE REGISTRATION STATEMENT

Item 3. INCORPORATION OF DOCUMENTS BY REFERENCE

Bath & Body Works, Inc. (the "Registrant") and the Plan hereby incorporate by reference into this Registration Statement the following documents filed with the Securities and Exchange Commission (the "Commission"):

(a) The Registrant's Annual Report on Form 10-K for the fiscal year ended January 30, 2021 (the "Registrant Annual Report"), filed with the Commission on March 19, 2021;

(b) The Annual Report on Form 11-K in respect of the Plan for the year ended December 31, 2020, filed with the Commission on June 29, 2021;

(c) All other reports filed pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), since the end of the fiscal year covered by the Registrant Annual Report; and

(d) The description of common stock, par value \$0.50 per share, of the Company contained in the Registrant's Form 8 Amendment to its Form 8-A, filed with the Commission on September 1, 1989 pursuant to the Exchange Act, including any amendment thereto or report filed for the purpose of updating such description.

All documents filed by the Registrant or the Plan pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of this Registration Statement and prior to the filing of a post-effective amendment that indicates that all securities offered hereby have been sold or which deregisters all securities then remaining unsold, are incorporated by reference in this Registration Statement and are a part hereof from the date of filing of such documents; except as to any portion of any future annual or quarterly report to stockholders or document or current report furnished under current Items 2.02 or 7.01 of Form 8-K that is not deemed filed under such provisions. Any statement contained in a document incorporated or deemed to be incorporated by reference herein shall be deemed to be modified or superseded for purposes of this Registration Statement to the extent that a statement contained herein or in any other subsequently filed document which also is or is deemed to be incorporated by reference herein modifies or supersedes such statement. Any such statement so modified or superseded shall not be deemed, except as so modified or superseded, to constitute a part of this Registration Statement.

Item 4. DESCRIPTION OF SECURITIES

Not applicable.

Item 5. INTERESTS OF NAMED EXPERTS AND COUNSEL

Not applicable.

Item 6. INDEMNIFICATION OF DIRECTORS AND OFFICERS

The Registrant is a Delaware corporation. Section 102(b)(7) of the Delaware General Corporation Law (the "DGCL") enables a corporation to eliminate or limit the personal liability of a director to the corporation or its stockholders for monetary damages for breach of the director's fiduciary duty, except:

- · for any breach of the director's duty of loyalty to the corporation or its stockholders;
- · for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law;
- pursuant to Section 174 of the DGCL (providing for liability of directors for unlawful payment of dividends or unlawful stock purchases or redemptions); or
- · for any transaction from which the director derived an improper personal benefit.

In accordance with Section 102(b)(7) of the DGCL, the Certificate of Incorporation of the Registrant includes a provision eliminating, to the fullest extent permitted by the DGCL, the liability of the Registrant's directors to the Registrant or its stockholders for monetary damages for breach of fiduciary duties as director.

Section 145(a) of the DGCL empowers a corporation to indemnify any present or former director, officer, employee or agent of the corporation, or any individual serving at the corporation's request as a director, officer, employee or agent of another organization, who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation), against expenses (including attorneys' fees), judgments, fines and amounts paid in settlement actually and reasonably incurred by the person in connection with such action, suit or proceeding provided that such person acted in good faith and in a manner he or she reasonably believed to be in, or not opposed to, the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his or her conduct was unlawful.

The DGCL provides that the indemnification described above shall not be deemed exclusive of any other indemnification that may be granted by a corporation pursuant to any by-law, disinterested directors' vote, stockholders' vote, agreement or otherwise.

The DGCL also provides corporations with the power to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation in a similar capacity for another corporation, partnership, joint venture, trust or other enterprise, against any liability asserted against him or her in any such capacity, or arising out of his or her status as such, whether or not the corporation would have the power to indemnify him or her against such liability as described above.

In accordance with Section 145(a) of the DGCL, the Registrant's Amended and Restated Bylaws provide that every person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit, or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that such person, or such person's testator or intestate, is or was serving as a director or officer of the Registrant or is or was serving at the request of the Registrant as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, or other enterprise, or as a member of any committee or similar body, shall be indemnified to the fullest extent permitted under the laws of Delaware against all expenses (including attorney's fees), judgments, penalties, fines and amounts paid in settlement actually and reasonably incurred by such person in connection with such action, suit or proceeding (including appeals) or the defense or settlement thereof or any claim, issue, or matter therein. Expenses incurred by a director or officer in defending such an action, suit or proceeding may be paid by the Registrant in advance of the final disposition of such action, suit or proceeding if such indemnification and/or payment is approved by the vote of the stockholders or of the disinterested directors, or is, in the opinion of independent legal counsel selected by the board of directors of the Registrant, to be made on behalf of an indemnitee who acted in good faith and in a manner such person reasonably believed to be in, or not opposed to, the best interests of the Registrant.

In addition, the Registrant's officers and directors are insured under an officers and directors liability insurance policy.

In addition, the Registrant has entered into separate indemnification agreements with certain of its current and former executive officers and directors. These indemnification agreements provide officers and directors with contractual rights to indemnification and advancement and reimbursement of expenses, to the fullest extent permitted under Delaware law, subject to certain exceptions contained in those agreements.

Item 7. EXEMPTION FROM REGISTRATION CLAIMED

Not applicable.

Item 8. EXHIBITS

Exhibit Number

i tumber	
4.1*	Amended and Restated Certificate of Incorporation of the Registrant (<u>incorporated by reference to Exhibit 3.1 to the Registrant's Current</u> <u>Report on Form 8-K, dated May 20, 2020</u>), as amended by the Certificate of Amendment of the Amended and Restated Certificate of Incorporation of the Registrant (<u>incorporated by reference to Exhibit 3.1 to the Registrant's Current Report on Form 8-K, dated August 3</u> , <u>2021</u>)
<u>4.2*</u>	Amended and Restated Bylaws of the Registrant (incorporated by reference to Exhibit 3.1 to the Registrant's Current Report on Form 8-K, dated January 28, 2022).
<u>4.3</u>	Bath & Body Works, Inc. 401(k) Savings and Retirement Plan, as amended and restated as of April 5, 2021 (filed herewith)
<u>4.4</u>	First Amendment to the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan dated as of July 28, 2021 (filed herewith)
<u>4.5</u>	Second Amendment to the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan dated as of December 29, 2021 (filed herewith)
<u>5.1</u>	Opinion of Davis Polk & Wardwell LLP (filed herewith)
<u>5.2</u>	Internal Revenue Service Determination Letter dated as of September 27, 2013 (filed herewith)
<u>15.1</u>	Letter of Awareness from Ernst & Young LLC (filed herewith)
<u>23.1</u>	Consent of Ernst & Young LLP (filed herewith)
<u>23.2</u>	Consent of Davis Polk & Wardwell LLP (included in Exhibit 5.1)
<u>24.1</u>	Power of Attorney of the Registrant (included on the signature pages of this Registration Statement)
<u>107.1</u>	Filing Fee Table (filed herewith)

*Incorporated herein by reference

Item 9. UNDERTAKINGS

- (a) The undersigned Registrant hereby undertakes:
 - (1) To file, during any period in which offers or sales are being made, a post-effective amendment to this Registration Statement:
 - (i) To include any prospectus required by Section 10(a)(3) of the Securities Act;

(ii) To reflect in the prospectus any facts or events arising after the effective date of this Registration Statement (or the most recent posteffective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in this Registration Statement. Notwithstanding the foregoing, any increase or decrease in volume of securities offered (if the total dollar value of securities offered would not exceed that which was registered) and any deviation from the low or high end of the estimated maximum offering range may be reflected in the form of prospectus filed with the Commission pursuant to Rule 424(b) if, in the aggregate, the changes in volume and price represent no more than a 20 percent change in the maximum aggregate offering price set forth in the "Calculation of Registration Fee" table in this registration statement; and

(iii) To include any material information with respect to the Plan not previously disclosed in this Registration Statement or any material change to such information in this Registration Statement;

provided, however, that paragraphs (a)(1)(i) and (a)(1)(ii) do not apply if the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed with or furnished to the Commission by the Registrant pursuant to Section 13 or Section 15(d) of the Exchange Act that are incorporated by reference in this Registration Statement.

(2) That, for the purpose of determining any liability under the Securities Act, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.

(3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.

(b) The undersigned Registrant hereby undertakes that, for purposes of determining any liability under the Securities Act, each filing of the Registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Exchange Act (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Exchange Act) that is incorporated by reference in this Registration Statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial *bona fide* offering thereof.

(c) Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers and controlling persons of the Registrant pursuant to the foregoing provisions, or otherwise, the Registrant has been advised that in the opinion of the Commission such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the Registrant of expenses incurred or paid by a director, officer, or controlling person of the Registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered hereunder, the Registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question of whether such indemnification by it is against public policy as expressed in the Securities Act and will be governed by the final adjudication of such issue.

SIGNATURES

Pursuant to the requirements of the Securities Act, the Registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing on Form S-8 and has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Columbus, State of Ohio, on the 10th day of February, 2022.

BATH & BODY WORKS, INC.

By: /s/ Wendy C. Arlin Name: Wendy C. Arlin Title: Executive Vice President and Chief Financial Officer

POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below, constitutes and appoints Andrew M. Meslow and Wendy C. Arlin and each of them, our true and lawful attorneys-in-fact and agents, with full power of substitution and resubstitution, to do any and all acts and things and execute, in the name of the undersigned, any and all instruments which said attorneys-in-fact and agents may deem necessary or advisable in order to enable Bath & Body Works, Inc. to comply with the Securities Act of 1933, as amended, and any requirements of the Securities and Exchange Commission in respect thereof, in connection with the filing with the Securities and Exchange Commission of one or more registration statements on Form S-8 under the Securities Act of 1933, as amended, including, specifically, but without limitation, power and authority to sign the name of the undersigned to any such registration statement, and any amendments to any such registration statement (including post-effective amendments), and to file the same with all exhibits thereto and other documents in connection therewith, with the Securities and Exchange Commission, to sign any and all applications, registration statements, notices or other documents necessary or advisable to comply with applicable state securities laws, and to file the same, together with other documents in connection therewith with the appropriate state securities authorities, granting unto said attorneys-in-fact and agents, and each of them, full power and authority to do and to perform each and every act and thing requisite or necessary to be done in and about the premises, as fully and to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents, and any of them, or their substitutes, may lawfully do or cause to be done by virtue hereof.

Pursuant to the requirements of the Securities Act of 1933, as amended, this Registration Statement has been signed by the following persons in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Andrew M. Meslow Andrew M. Meslow	Director and Chief Executive Officer (principal executive officer)	February 10, 2022
/s/ Wendy C. Arlin Wendy C. Arlin	Executive Vice President and Chief Financial Officer (principal financial and principal accounting officer)	February 10, 2022
/s/ Sarah E. Nash Sarah E. Nash	Director and Chair	February 10, 2022
/s/ Patricia S. Bellinger Patricia S. Bellinger	Director	February 10, 2022
/s/ Francis A. Hondal Francis A. Hondal	Director	February 10, 2022
/s/ Danielle M. Lee Danielle M. Lee	Director	February 10, 2022
/s/ Michael G. Morris Michael G. Morris	Director	February 10, 2022
/s/ Robert H. Schottenstein Robert H. Schottenstein	Director	February 10, 2022
/s/ Stephen D. Steinour Stephen D. Steinour	Director	February 10, 2022
/s/ James K. Symancyk James K. Symancyk	Director	February 10, 2022

Pursuant to the requirements of the Securities Act of 1933, as amended, the undersigned administrator of the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Columbus, State of Ohio, on the 10th day of February, 2022.

BATH & BODY WORKS, INC. 401(k) SAVINGS AND RETIREMENT PLAN

By: /s/ Deon N. Riley

Name:Deon N. RileyTitle:Chief Human Resources Officer,
Bath & Body Works, Inc.

<u>L BRANDS, INC.</u>

401(K) SAVINGS AND RETIREMENT PLAN

(As Amended and Restated Effective as of April 5, 2021)

		Page
Article I	INTRODUCTION	1
1.1.	Adoption and Effective Date	1
1.2.	Purpose	1
1.3.	Applicability	1
Article II	DEFINITIONS	2
2.1.	Account	2
2.2.	ADS Transferred Associate	2
2.3.	ADS Transition Agreement	2
2.4.	ADS Transfer Date	2
2.5.	Affiliate	2
2.6.	Associate	2
2.7.	Beneficiary	3
2.8.	Board of Directors	3
2.9.	Break in Service	3
2.10.	Catch-Up Account	3
2.11.	Catch-Up Contributions	3
2.12.	Code	3
	Compensation	3
2.14.	Compensation Committee	4
2.15.	Effective Restatement Date	4
2.16.	Eligible Associate	4
2.17.	Eligibility Computation Period	4
2.18.	Employer	5
2.19.	Employer Contribution Enrollment Date	5
2.20.	Employer Securities	5
2.21.	Employer Securities Contribution Fund	5
2.22.	Enrollment and Change Designation	5
2.23.	Enrollment Period	5
2.24.	ERISA	5
	Full-Time Associate	5
2.26.	Highly Compensated Associate	5
	Hour of Service	5
2.28.	Investment Funds	6
2.29.	Key Associate	6
2.30.	Leased Employee	6
2.31.	Limitation Year	6
2.32.	L Brands	6
2.33.	Matching Account	6
2.34.	Matching Contributions	6
2.35.	Non-Highly Compensated Associate	6
2.36.	Non-Key Associate	6

2.3.8 Patricipant 6 2.3.8 Patricipant 7 2.4.9 Pan Sponsor 7 2.4.1 Pan Sponsor 7 2.4.2 Post-Tax Savings Account 7 2.4.3 Post-Tax Savings Contributions 7 2.4.4 Pre-Tax Savings Contributions 7 2.4.4 Pre-Tax Savings Contributions 7 2.4.5 Pre-Tax Savings Contributions 7 2.4.6 Retirement Contribution 7 2.4.7 Retirement Contribution 7 2.4.8 Retirement Contribution 7 2.4.8 Retirement Contribution 7 2.4.8 Retirement Contribution 7 2.5.8 Roht Conversion Announts 7 2.5.8 Roht Conversion Announts 8 2.5.8 Roht Savings Account 8 2.5.9 <			
2.3.Plan72.4.Plan Year72.4.1Plan Year72.4.2Post-Tax Savings Account72.4.3Post-Tax Savings Contributions72.4.4Pre-Tax Savings Contributions72.4.4Pre-Tax Savings Contributions72.4.4Pre-Tax Savings Contributions72.4.5Pre-Tax Savings Contributions72.4.6Retirement Account72.4.7Retirement Plan Committee72.4.8Retirement Plan Committee72.5.9Rollover Account72.5.1Roth Conversion Amounts72.5.2Roth Conversion Subaccount82.5.3Roth Sollover Account82.5.4Roth Conversion Subaccount82.5.5Roth Savings Contribution82.5.6Roth Savings Contribution82.5.7Savings Contribution Enrollment Date82.5.8Roth Savings Contribution Enrollment Date82.5.9Separation Date82.6.1Taxable Wage Base92.6.2Trust Eud or Trust92.6.3Trust Base92.6.4Trust Eud or Trust92.6.5Trust Eud or Trust92.6.6Trust Eud or Trust92.6.6Trust Eud or Trust92.6.7Trust Eud or Trust92.6.6Trust Eud or Trust92.6.7Trust Eud or Trust92.6	2.37.	Normal Retirement Date	6
2.40.Plan Sponsor72.41.Plan Year72.42.Post-Tax Savings Contributions72.43.Post-Tax Savings Contributions72.44.Pre-Tax Savings Contributions72.45.Pre-Tax Savings Contribution72.46.Retirement Contribution72.47.Retirement Contribution72.48.Retirement Contribution72.49.Rollover Contribution72.40.Rollover Contribution72.51.Roth Conversion Amounts72.52.Roth Conversion Amounts82.53.Roth Rollover Account82.54.Roth Rollover Account82.55.Roth Rollover Contribution82.55.Roth Rollover Contribution82.55.Roth Savings Account82.55.Roth Savings Contributions82.55.Roth Savings Contribution82.55.Roth Savings Contribution82.55.Roth Savings Contribution82.55.Roth Savings Contribution82.55.Roth Savings Contribution82.55.Roth Savings Contribution82.56.Roth Savings Contribution82.57.Roth Savings Contribution82.58.Roth Savings Contribution92.60.Yean G92.61.Traste92.62.Traste92.63.Yean G92.	2.38.	Participant	6
2.41Plan Yar72.42Post-Tax Savings Account72.43Post-Tax Savings Contributions72.44Pre-Tax Savings Contributions72.45Pre-Tax Savings Contributions72.46Refirement Account72.47Refirement Contribution72.48Refirement Contribution72.49Rollover Account72.40Rollover Contribution72.41Rollover Contribution72.52Rollover Contribution72.53Roll Conversion Subaccount82.54Rollover Account82.55Roll Rollover Contribution82.56Rollover Account82.57Savings Contribution Eurollment Date82.58Roll Savings Contribution Eurollment Date82.59Separation Date82.50Souse82.51Tust Agreement92.64Trust Agreement92.65Trust Para92.65Trust Para92.66Trust Paration Date92.67Valation Date92.68Vaer of Eligibility Service92.69Vaer of Eligibility Service92.60Trust Parate133.70Change in Strust133.71Rollation of Ligible Associate133.72Contributions133.73Contributions133.74	2.39.	Plan	
2.42Post-Tac Savings Account72.43Post-Tac Savings Contributions72.44Pre-Tax Savings Contributions72.45Pre-Tax Savings Contributions72.46Reitement Contribution72.47Reitement Contribution72.48Reitement Contribution72.49Rollover Contribution72.40Rollover Contribution72.51Roth Conversion Amounts72.52Roth Conversion Amounts82.53Roth Rollover Contribution82.54Roth Rollover Contribution82.55Roth Savings Account82.56Roth Savings Contribution82.57Savings Contribution82.58Roth Savings Contribution82.59Separation Date82.50Roth Savings Contribution82.55Roth Savings Contribution82.56Roth Savings Contribution82.57Savings Contribution82.58Self-Managed Brokerage Accounts82.59Separation Date82.60Spruse92.61Trask Agreement92.65Trusk Agreement92.66Trusk Agreement92.67Valuation Date92.68Year of Eligibility Service102.61Anticle HIPARTICIPATION102.74Aluison of Ineligible Associate13 <trt< td=""><td>2.40.</td><td>Plan Sponsor</td><td></td></trt<>	2.40.	Plan Sponsor	
2.43Post-Tax Savings Contributions72.44Pre-Tax Savings Contributions72.45Pre-Tax Savings Contributions72.46Retirement Account72.47Retirement Contribution72.48Retirement Contribution72.48Rollover Account72.49Rollover Contribution72.51Roth Conversion Subaccount82.52Roth Rollover Contribution82.53Roth Rollover Contribution82.54Roth Rollover Contribution82.55Roth Savings Contributions82.56Roth Savings Contributions82.57Savings Contributions82.58Roth Savings Contributions82.59Roth Savings Contributions82.50Roth Savings Contributions82.51Roth Savings Contributions82.55Roth Savings Contributions82.56Roth Savings Contributions82.57Savings Contributions82.58Self-Managel Brokerage Accounts82.59Separation Date82.60Spouse82.61Trust Agreement92.65Trust egreement92.65Trust egreement92.66Trust egreement92.67Valuation Date103.7Lingibility and Participation103.8Conage in Staus123.9<	2.41.	Plan Year	
2.44Pre-Tax Savings Countilutions72.45Pre-Tax Savings Countibutions72.46Retirement Account72.47Retirement Committee72.48Retirement Committee72.49Rollover Account72.50Rohlover Contribution72.51Roth Conversion Amounts72.52Roth Conversion Subaccount82.53Roth Rollover Account82.54Roth Rollover Contribution82.55Roth Savings Account82.55Roth Savings Contributions82.55Roth Savings Contributions82.56Roth Savings Contribution Enrollment Date82.57Savings Contribution Enrollment Date82.58Separation Date82.60Spouse82.61Travable Wage Base82.62Top-Heavy Plan92.64Trust Agreement92.65Trustee92.66Trust Fund or Trust92.67Vating Dare92.68Year of Vesting Service92.69Year of Eligbility Service92.60Year of Eligbility Service92.61Trust Magreement103.7Anage in Status123.8Self-Ulation103.1Lingsion of Eligble Associate133.5Inclusion of Ineligible Associate133.5Inclusion of Ineligibl	2.42.	Post-Tax Savings Account	
2.45Pre-Tax Saving's Contributions72.46Retirement Account72.47Retirement Contribution72.48Retirement Contribution72.48Retirement Contribution72.49Rollover Account72.50Rollover Account72.51Roth Conversion Subaccount82.53Roth Rollover Account82.54Roth Rollover Account82.55Roth Rollover Account82.56Roth Rollover Account82.57Savings Contribution82.58Self-Managed Brokerage Accounts82.59Separation Date82.60Spouse82.61Taxable Wage Base82.62Top-Heavy Plan82.63Total and Permanent Disability92.64Trust Agreement92.65Trust Querement92.66Trust Pand or Trust92.67Vau of Date92.68Year of Eligibility Service92.69Year of Vesting Service103.1Eligibility and Participation123.2Kalver of Participation123.3Change in Staus133.4Charley Associate133.5Inclusion of Heiligible Associate133.6Cestion of Participation by ADS Transferred Associates133.6Pre-Tax and Roth Savings Contributions133.7 <td></td> <td></td> <td></td>			
2.46.Retirement Account72.47.Retirement Account72.48.Retirement Plan Committee72.48.Retirement Plan Committee72.50.Rollover Account72.51.Roth Conversion Amounts72.52.Roth Conversion Subaccount82.53.Roth Rollover Account82.54.Roth Rollover Contribution82.55.Roth Savings Account82.55.Roth Savings Account82.55.Roth Savings Account82.55.Roth Savings Account82.55.Self-Managed Brokerage Accounts82.65.Sourge Contribution Enrollment Date82.66.Traxable Wage Base82.61.Traxable Wage Base82.62.Total and Permanent Disability92.64.Trust Agreement92.65.Trust Agreement92.66.Trust Parement92.67.Year of Eligibility Service92.68.Year of Eligibility Service92.69.Year of Eligibility Service92.60.Soursion of Eligible Associate133.61.Change in Status123.61.Change in Status133.61.Contributions133.61.Per-Tax and Roth Savings Contributions133.61.Per-Tax and Roth Savings Contributions133.62.Per-Tax and Roth Savings Contributions13 <td>2.44.</td> <td>Pre-Tax Savings Account</td> <td>7</td>	2.44.	Pre-Tax Savings Account	7
2.47.Retirement Contribution72.48.Retirement Plan Committee72.49.Rollover Account72.50.Rollover Contribution72.51.Roth Conversion Amounts72.52.Roth Conversion Subaccount82.53.Roth Roth Roth Conversion Subaccount82.54.Roth Rollover Contribution82.55.Roth Rollover Contribution82.55.Roth Rollover Contribution82.55.Roth Savings Contribution Enrollment Date82.56.Roth Savings Contribution Enrollment Date82.55.Spouse82.56.Spouse82.57.Saving Sourtibution Enrollment Date82.58.Spouse82.59.Spearation Date82.60.Trust Agreement92.61.Trust Agreement92.65.Trust Agreement92.66.Trust Agreement92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service103.1.Eligibility Service123.3.Change in Status123.4.Change in Status133.5.Inclusion of Ineligible Associate133.6.Cestion of Participation133.6.Cestion of Participation by ADS Transferred Associates133.6.Cestion of Participation by ADS Transferred Associates13 </td <td>2.45.</td> <td>Pre-Tax Savings Contributions</td> <td></td>	2.45.	Pre-Tax Savings Contributions	
2.48.Retirement Plan Committee72.49.Rollover Account72.50.Roth conversion Amounts72.51.Roth Conversion Amounts82.53.Roth Conversion Subaccount82.53.Roth Rollover Account82.54.Roth Rollover Account82.55.Roth Savings Account82.55.Roth Savings Account82.55.Roth Savings Account82.55.Roth Savings Accounts82.55.Separation Date82.50.Spouse82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Trust Eud or Trust92.64.Trust Pund or Trust92.65.Trust Eud or Trust92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service92.69.Year of Vesting Service103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Ichange in Staus133.4.Constion of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Creation of Participation by ADS Transferred Associates133.6.Creation of Participation by ADS Transferred Associates133.6.Contributions	2.46.	Retirement Account	
2.49.Rollover Account72.50.Rollover Contribution72.51.Roth Conversion Subaccount82.53.Roth Rollover Account82.54.Roth Solvings Account82.55.Roth Savings Contributions82.55.Roth Savings Contributions82.55.Roth Savings Contributions82.55.Schot Savings Contributions82.56.Travable Wage Base82.61.Travable Wage Base92.62.Top-Heavy Plan82.63.Trust Agreement92.64.Trust Stave92.65.Trust Eud or Trust92.66.Trust Eud or Trust92.67.Valuation Date92.68.Year of Vesting Service92.69.Year of Vesting Service92.60.Trust Eud or Trust103.1.Eligibility and Participation123.2.Waiver of Participation123.3.Change in Status133.40.IPAITCIPATION133.41.Retirement Contributions133.51.Inclusion of Eligible Associate13 </td <td>2.47.</td> <td>Retirement Contribution</td> <td></td>	2.47.	Retirement Contribution	
2.50.Rollover Contribution72.51.Roth Conversion Subaccount82.53.Roth Conversion Subaccount82.54.Roth Rollover Account82.55.Roth Savings Account82.56.Roth Savings Contribution82.57.Savings Contribution82.58.Self-Managed Brokerage Accounts82.59.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trust Eurd or Trust92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Tust Service103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Change in Stratus133.4.Omission of Eligible Associate133.5.Inclusion of Participation by ADS Transferred Associates133.6.VONTRIBUTIONS134.1.Retirement Contributions134.2.Yer-Tax and Roth Savings Contributions15	2.48.	Retirement Plan Committee	
2.51.Roth Conversion Amounts72.52.Roth Conversion Subaccount82.53.Roth Rollover Account82.54.Roth Savings Account82.55.Roth Savings Contribution82.55.Roth Savings Contributions82.55.Roth Savings Contributions82.55.Separation Date82.55.Separation Date82.55.Separation Date82.55.Separation Date82.55.Separation Date82.55.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trustee92.66.Trust Pund or Trust92.66.Trust Pund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service10Article IIIPARTICIPATION103.1.Eligibility and Participation123.28.Guage Ibaseciate133.5.Inclusion of Eligible Associate133.6.Cessation of Participation by ADS Transferred Associates133.7.Inclusion of Ibasing Associate133.8.Cessation of Participation by ADS Transferred Associates133.4.Pre-Tax and Roth Savings Co	2.49.	Rollover Account	
2.52.Roth Conversion Subaccount82.53.Roth Rollover Account82.54.Roth Rollover Contribution82.55.Roth Savings Account82.55.Roth Savings Account82.55.Savings Contribution Enrollment Date82.55.Savings Contribution Enrollment Date82.55.Self-Managed Brokerage Accounts82.56.Spouse82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trust Fund or Trust92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Tustify Service92.69.Year of Tustify Service92.61.Indicipation103.1.Eligibility and Participation123.2.Waiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Participation by ADS Transferred Associates133.6.Cesation of Participation by ADS Transferred Associates133.6.Cesation of Participation by ADS Transferred Associates133.6.Pre-Tax and Roth Savings Contributions134.2.Pre-Tax and Roth Savings Contributions13<	2.50.	Rollover Contribution	7
2.53.Roth Rollover Account82.54.Roth Rollover Account82.55.Roth Savings Account82.55.Roth Savings Contributions82.57.Savings Contribution Enrollment Date82.58.Self-Managed Brokerage Accounts82.59.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trust equerement92.66.Trust fund or Trust92.66.Year of Eligibility Service92.66.Year of Vesting Service92.66.Year of Vesting Service92.67.Valuation Date92.68.Year of Participation103.1.Eligibility and Participation103.2.Uraiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Costraticipation by ADS Transferred Associates133.6.Contributions134.7.VONTRIBUTIONS134.7.Kerrement Contributions134.2.Pre-Tax and Roth Savings Contributions13	2.51.	Roth Conversion Amounts	7
254.Roth Rollover Contribution82.55.Roth Savings Account82.56.Roth Savings Contributions82.57.Savings Contribution Enrollment Date82.58.Self-Managed Brokerage Accounts82.59.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trust Querement92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Vesting Service92.69.Year of Vesting Service92.69.Year of Uesting Service92.69.Vesting Service92.69.Vesting Service92.69.Vesting Service92.69.Vesting Service92.69.Vesting Service92.69.Vesting Service92.69.Vesting Service92.69.Vesting Service92.69.Vesting Service103.10.Eligibility and Participation103.20.Nange of Eligibility Service133.30.Change in Status133.40.Omission of Eligible Associate133.50.Icclusion of Iligible Associate133.60.Cessation of Participation by ADS Transferred Associates133.	2.52.	Roth Conversion Subaccount	8
2.55.Roth Savings Account82.55.Roth Savings Contribution Enrollment Date82.57.Savings Contribution Enrollment Date82.58.Self-Managed Brokerage Accounts82.59.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Truste92.66.Trust Agreement92.67.Valuation Date92.68.Year of Vesting Service92.69.Year of Vesting Service92.69.Year of Vesting Service92.61.Status10Article IIIPARTICIPATION103.2.Waiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Participation by ADS Transferred Associates133.6.Cessation of Participation by ADS Transferred Associates133.6.Cessation of Participation by ADS Transferred Associates133.6.Reitrem Contributions134.2.Pre-Tax and Roth Savings Contributions15	2.53.	Roth Rollover Account	8
2.56.Roth Savings Contributions82.57.Savings Contribution Enrollment Date82.58.Self-Managed Brokerage Accounts82.59.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trust Ragreement92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service10Article IIIPARTICIPATION103.1.Eligibility and Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates133.6.Cessation of Participation by ADS Transferred Associates133.7.Per-Tax and Roth			8
2.57.Savings Contribution Enrollment Date82.58.Self-Managed Brokerage Accounts82.59.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trustee92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service10Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Chauge in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates133.6.Cestation of Participation by ADS Transferred Associates133.7.Per-Tax and Roth Savings Cont	2.55.	Roth Savings Account	8
2.58.Self-Managed Brokerage Accounts82.59.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Truste Fund or Trust92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service10Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions13	2.56.	Roth Savings Contributions	8
2.59.Separation Date82.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trust Germent92.66.Trust Und or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service92.61.Iligibility and Participation103.1.Eligibility and Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions15	2.57.	Savings Contribution Enrollment Date	8
2.60.Spouse82.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trustee92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Eligibility Service10Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Change in Status123.3.Change in Status133.5.Inclusion of Eligible Associate133.6.Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions15	2.58.	Self-Managed Brokerage Accounts	8
2.61.Taxable Wage Base82.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Trust Agreement92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Eligibility Service92.69.Year of Eligibility Service10Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Change in Status123.3.Change in Status133.5.Inclusion of Eligible Associate133.6.Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions15	2.59.	Separation Date	
2.62.Top-Heavy Plan82.63.Total and Permanent Disability92.64.Trust Agreement92.65.Truste Agreement92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service92.61.Ligibility and Participation103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions13			
2.63.Total and Permanent Disability92.64.Trust Agreement92.65.Truste92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service10Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions15			8
2.64Trust Agreement92.65Trustee92.66Trust Fund or Trust92.67Valuation Date92.68Year of Eligibility Service92.69Year of Vesting Service10Article IIIPARTICIPATION103.1Eligibility and Participation103.2Waiver of Participation123.3Change in Status123.4Omission of Eligible Associate133.5Inclusion of Ineligible Associate133.6Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1Retirement Contributions134.2Pre-Tax and Roth Savings Contributions13	2.62.	Top-Heavy Plan	8
2.65.Trustee92.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service10Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions13			
2.66.Trust Fund or Trust92.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service10Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Vaiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions13	2.64.	Trust Agreement	
2.67.Valuation Date92.68.Year of Eligibility Service92.69.Year of Vesting Service10Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions15	2.65.	Trustee	9
2.68. 2.69. Year of Vesting Service9Article IIIPARTICIPATION3.1.Eligibility and Participation3.2.Waiver of Participation3.3.Change in Status3.4.Omission of Eligible Associate3.5.Inclusion of Ineligible Associate3.6.Cessation of Participation by ADS Transferred Associates3.7.CONTRIBUTIONS4.1.Retirement Contributions4.2.Pre-Tax and Roth Savings Contributions	2.66.	Trust Fund or Trust	9
2.69. Year of Vesting Service10Article IIIPARTICIPATION103.1. Eligibility and Participation103.2. Waiver of Participation123.3. Change in Status123.4. Omission of Eligible Associate133.5. Inclusion of Ineligible Associate133.6. Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15	2.67.	Valuation Date	9
Article IIIPARTICIPATION103.1.Eligibility and Participation103.2.Waiver of Participation123.3.Change in Status123.4.Omission of Eligible Associate133.5.Inclusion of Ineligible Associate133.6.Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1.Retirement Contributions134.2.Pre-Tax and Roth Savings Contributions13	2.68.	Year of Eligibility Service	9
3.1. Eligibility and Participation103.2. Waiver of Participation123.3. Change in Status123.4. Omission of Eligible Associate133.5. Inclusion of Ineligible Associate133.6. Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15	2.69.	Year of Vesting Service	10
3.1. Eligibility and Participation103.2. Waiver of Participation123.3. Change in Status123.4. Omission of Eligible Associate133.5. Inclusion of Ineligible Associate133.6. Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15	Article III	PARTICIPATION	10
3.2. Waiver of Participation123.3. Change in Status123.4. Omission of Eligible Associate133.5. Inclusion of Ineligible Associate133.6. Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15			
3.3. Change in Status123.4. Omission of Eligible Associate133.5. Inclusion of Ineligible Associate133.6. Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15			
3.4. Omission of Eligible Associate133.5. Inclusion of Ineligible Associate133.6. Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15			
3.5. Inclusion of Ineligible Associate133.6. Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15			
3.6. Cessation of Participation by ADS Transferred Associates13Article IVCONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15			
Article IV CONTRIBUTIONS134.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15		-	
4.1. Retirement Contributions134.2. Pre-Tax and Roth Savings Contributions15	3.6.	Cessation of Participation by ADS Transferred Associates	13
4.2. Pre-Tax and Roth Savings Contributions 15	Article IV	CONTRIBUTIONS	13
4.2. Pre-Tax and Roth Savings Contributions 15	4.1.	Retirement Contributions	13
		-	16

ii

4.4	Pallavar Contributions	17
	Rollover Contributions	17
	Timing of Contributions	18
	Annual Additions and Limitations	18
	Fail-Safe Allocations of Retirement Contributions	19
	Forfeitures	20
	Exclusive Benefit; Refund of Contributions	20
	Catch-Up Contribution	20
	Minimum Top-Heavy Contributions	21
4.12.	Nondiscrimination Test for Disaggregated Pre-Tax and Roth Savings Contributions	21
4.13.	Roth Conversion Amounts	24
Article V	INVESTMENT OF THE TRUST FUND	24
5.1.	Investment Direction	24
5.2.	Absence of Investment Direction	25
	Investment Funds	25
	Investment in Employer Securities	25
	Voting Employer Securities	26
	Tender Offers	26
	Investment Managers	20
	Valuations	27
		27
5.9.	Managed Account Service	27
Article VI	ALLOCATIONS TO PARTICIPANT ACCOUNTS	27
6.1.	Participant Accounts	27
6.2.	Credits to and Charges Against Accounts	28
6.3.	Expenses	28
6.4.	Reimbursement of Trust Fund	28
Article VII	VESTING AND SEPARATION FROM SERVICE	28
7.1.	Vested Percentage	28
	Forfeiture and Restoration	30
	Effect of Breaks in Service	31
	Amendments to Vesting Schedule	31
		_
Afficie v III	BENEFITS	32
8.1.	Benefit Payments after Separation Date	32
8.2.	Retirement Benefit	34
8.3.	Death Benefit	36
8.4.	Beneficiary Designation	36
	In-Service Withdrawals	37
	Post Distribution Credits	39
	Lost Participants	40
Article IX	THE RETIREMENT PLAN COMMITTEE	40
Q 1	Appointment and Tenure	40
9.2.	Meetings; Majority Rule	41

iii

9.3.	Delegation	41
9.4.	Appointment of Trustee	41
9.5.	Investment Authority	41
	Reporting and Disclosure	41
	Construction of the Plan	42
	Engagement of Assistants and Advisors	42
	Bonding	42
	Compensation	42
	Indemnification of the Retirement Plan Committee	42
9.11.		42
Article X	ALLOCATION OF AUTHORITY AND RESPONSIBILITIES	43
10.1.	Authority and Responsibilities of the Plan Sponsor	43
	Authority and Responsibilities of the Retirement Plan Committee	43
	Authority and Responsibilities of the Trustee	43
	Limitations on Obligations	43
10.4.		-0
Article XI	CLAIMS PROCEDURES	43
11.1.	Application for Benefits	43
	Appeals of Denied Claims for Benefits	44
	Review of Decision	44
11.0.		
Article XII	AMENDMENT, TERMINATION, MERGERS AND CONSOLIDATIONS	45
12.1.	Amendment	45
12.2.	Termination	45
	Permanent Discontinuance of Contributions	45
	Suspension of Employer Contributions	46
	Mergers and Consolidations of Plans	46
	Transfers of Assets to or from the Plan	46
12.0.		40
Article XIII	PARTICIPATING EMPLOYERS	46
13.1.	Adoption by Other Entities	46
	Requirements of Participating Employers	46
	Designation of Agent	47
	Discontinuance of Participation	47
	Retirement Plan Committee's Authority	47
	MISCELLANEOUS PROVISIONS	47
	Nonalienation of Benefits	47
	No Contract of Employment	48
	Severability	48
	Successors	48
14.5.	Captions	48
14.6.	Gender and Number	48
14.7.	Controlling Law	48
14.8.	Title to Assets	48
14.9.	Payments to Minors, Etc	48

iv

14.10.	Risk to Participants	49
14.11.	Entire Agreement; Successors	49
14.12.	Approval by Internal Revenue Service	49
14.13.	Electronic and Telephonic Elections	49
14.14.	Special Provisions for Participants Who Enter the Armed Forces	49
14.15.	Correction of Mistakes	49
14.16.	Restriction on Venue and Limitations on Actions	49

L BRANDS, INC. 401(K) SAVINGS AND RETIREMENT PLAN

(As Amended and Restated Effective as of April 5, 2021)

ARTICLE I

INTRODUCTION

1.1. <u>Adoption and Effective Date</u>. The Limited Stores, Inc. adopted The Limited Stores, Inc. Savings and Retirement Plan (the "Plan") effective as of August 1, 1971. Effective as of January 1, 1992, profit sharing plans sponsored by certain affiliates of The Limited, Inc. were merged into the Plan. Also effective as of January 1, 1992, Limited Service Corporation, a Delaware corporation (the "Plan Sponsor"), assumed sponsorship of the Plan which was amended, restated and renamed The Limited, Inc. Savings and Retirement Plan. The Plan was subsequently amended and restated in its entirety effective as January 1, 2002, April 1, 2002, January 1, 2005 January 1, 2007, January 1, 2009 and January 1, 2010. Pursuant to the April 1, 2002 Plan restatement, the name of the Plan was changed to the Limited Brands, Inc. Savings and Retirement Plan. The Plan was further amended and restated, effective October 1, 2011, to make certain changes to the design of the Plan and to reflect the change in the name of the Plan Sponsor from Limited Service Company, LLC. The Plan was subsequently amended, effective December 2, 2013, to reflect the change in the name of the Plan Sponsor from Limited Brands Service Company, LLC to L Brands Service Company, LLC and to reflect the change in the name of the Plan from the Limited Brands, Inc. 401(k) Savings and Retirement Plan to the L Brands, Inc. 401(k) Savings and Retirement Plan. The Plan is hereby amended and restated, effective April 5, 2021 (except as otherwise stated herein), to incorporate prior amendments, add Roth contributions and in-plan Roth conversions, add a twenty-five percent limit on newly-elected investments into the Employer Securities Contribution Fund effective as of June 24, 2021, permit partial distributions following a Participant's termination of employment, modify hardship distribution provisions in accordance with the Bipartisan Budget Act of 2018, allow for the use of managed account services, and to make certain other changes.

1.2. <u>Purpose</u>. The Plan Sponsor adopts this amendment and restatement of the Plan in order to continue to provide retirement benefits for the Eligible Associates of participating Employers, to continue to provide Eligible Associates with the opportunity to reduce their current income for Federal income tax purposes while saving for retirement and to give Eligible Associates a means for acquiring shares of common stock of L Brands and thereby sharing in the growth and success of L Brands and its Affiliates. The Plan Sponsor intends that the Plan, as amended and restated herein, shall continue to qualify under Sections 401(a) and 401(k) of the Code, as amended, and shall meet the requirements of ERISA, including Section 404(c) of ERISA regarding participant-directed investments, and that the Trust established in connection with the Plan shall continue to be exempt from taxation under Section 501(a) of the Code.

1.3. <u>Applicability</u>. This amendment and restatement of the Plan is generally effective as of April 5, 2021 (the "Effective Restatement Date"). However, certain provisions of this amendment and restatement of the Plan are effective as of some other date. The provisions of

this amendment and restatement of the Plan that are effective prior to the Effective Restatement Date shall be deemed to amend the corresponding provisions of the Plan as in effect before this amendment and restatement and all amendments thereto. Events occurring before the applicable effective date of any provision of this amendment and restatement of the Plan shall be governed by the applicable provision of the Plan in effect on the date of the event. The rights and benefits of Associates whose participation terminated prior to the Effective Restatement Date shall be determined in accordance with the Plan as in effect on the day such Associates ceased participation in the Plan.

ARTICLE II

DEFINITIONS

The following terms shall have the meanings assigned in this ARTICLE II, which shall be equally applicable to the singular and plural forms of such terms, unless the context requires otherwise, when used in the Plan.

2.1. <u>Account</u> means the account maintained for a Participant under the Plan. A Participant's Account shall consist of his or her Retirement, Pre-Tax Savings, Catch-Up, Post-Tax Savings, Matching, Rollover, Roth Savings, and Roth Rollover Accounts.

2.2. <u>ADS Transferred Associate</u> means a former associate of L Brands Direct Fulfillment, Inc. who is a "Transferred Employee" as defined in Section 2.01 of the ADS Transition Agreement.

2.3. <u>ADS Transition Agreement</u> means that certain Employee Transition Agreement dated as of June 12, 2012, by and between ADS Alliance Data Systems, Inc. and L Brands Direct Fulfillment, Inc.

2.4. <u>ADS Transfer Date</u> means the Transfer Date (as defined in Section 1.09 of the ADS Transition Agreement) of an ADS Transferred Associate.

2.5. <u>Affiliate</u> means: (i) any corporation which is a member of a controlled group of corporations, as defined in Section 414(b) of the Code, of which L Brands is a member; (ii) any other trade or business (whether or not incorporated) which is under common control, as defined in Section 414(c) of the Code, with L Brands; (iii) any business which is a member of an affiliated service group, as defined in Section 414(m) of the Code, of which L Brands is a member; and (iv) any other entity required to be aggregated with L Brands pursuant to regulations issued under Section 414(o) of the Code.

2.6. <u>Associate</u> means any person employed by the Employer on its domestic U.S. payroll other than a person who is (i) described in Section 410(b)(3)(A) of the Code and with respect to whom inclusion in the Plan has not been provided for in the collective bargaining agreement setting forth his or her terms and conditions of employment, (ii) described in Section 410(b)(3)(C) of the Code, (iii) a Leased Employee, (iv) an independent contractor, or (v) employed by Victoria's Secret Stores Puerto Rico, LLC, which is an Affiliate of the Employer. If an individual is excluded from participation as an independent contractor and is later reclassified as a common law employee for wage and hour purposes, such individual shall be eligible to

participate only as of the date of his or her reclassification prospectively and only upon the completion of the eligibility requirements set forth herein.

2.7. <u>Beneficiary</u> means the person or persons designated by a Participant in accordance with Section 8.4 as his or her beneficiary or beneficiaries under the Plan in the event of the Participant's death.

2.8. <u>Board of Directors means the Board of Directors of L Brands.</u>

2.9. <u>Break in Service</u> means a Plan Year in which an Associate does not complete at least 500 Hours of Service. If an Associate is absent from work for any period by reason of vacation, holiday, illness, incapacity (including disability), jury duty, military duty or leave of absence or classified by the Employer as furloughed without pay, then for purposes of determining whether a Break in Service has occurred, such Associate shall be credited with the Hours of Service which otherwise normally would have been credited to such Associate, or, if the Retirement Plan Committee is unable to determine the number of such Hours of Service, eight Hours of Service for each day of such absence, not to exceed 501 Hours. Additionally, for purposes of determining whether an Associate has incurred a Break in Service, the Associate shall be credited with up to 501 Hours of Service for any period of absence from work for maternity or paternity reasons resulting from: (a) pregnancy of the individual; (b) birth of a child of the individual; (c) placement of a child with the individual in connection with the adoption of the child by the individual; or (d) caring for the child by the individual for a period beginning immediately after the birth or placement. The Hours of Service credited to an Associate under this definition shall be treated as Hours of Service in the Plan Year in which the absence from work begins, if the Associate would be prevented from incurring a Break in Service in such year solely because of such Hours of Service or, in any other case, in the immediately following Plan Year. The Retirement Plan Committee may require that the Associate certify and/or supply documentation that his or her absence is for one of the permitted reasons and the number of days for which there was such an absence.

2.10. Catch-Up Account means the portion of the Account of a Participant consisting of Catch-Up Contributions, as adjusted under the Plan.

2.11. <u>Catch-Up Contributions</u> means the amount contributed by an Employer pursuant to an eligible Participant's election and treated as a Catch-Up Contribution in accordance with Section 4.10 of the Plan.

2.12. Code means the Internal Revenue Code of 1986, as amended, and the regulations issued thereunder.

2.13. <u>Compensation</u> means amounts received by an Eligible Associate from an Employer while the Eligible Associate is a Participant which are base salary or wages, overtime payments, vacation, holiday and sick pay, short-term disability pay, bonuses (other than sign-on bonuses or retention bonuses), commissions, contest earnings or any other direct current compensation which is required to be reflected on the Participant's Form W-2 for the Plan Year, without giving effect to any reduction of compensation resulting from an Enrollment and Change Designation or a salary reduction arrangement pursuant to Section 125 of the Code or qualified

transportation fringe benefits, as described in Section 132(f)(4) of the Code, but shall not include Employer contributions to Social Security, other Employer or Associate contributions to this or any other deferred compensation plan or program, severance pay, stock options, long term disability income payments, relocation expense reimbursement, or the value of any other fringe benefits (such as non-cash contest earnings) provided at the expense of the Employer. To the extent permitted by, and subject to the provisions of, the Heroes Earnings Assistance and Relief Tax Act of 2008 and Section 3401(h) of the Code, Compensation shall include any differential wage payments, as defined in Section 3401(h)(2) of the Code.

The annual Compensation of each Participant taken into account under the Plan shall not exceed the limit established under Section 401(a)(17)(A) of the Code (\$ 290,000 for 2021), which shall be adjusted for cost-of-living increases in accordance with Section 401(a)(17)(B) of the Code, as amended by the Economic Growth and Tax Relief Reconciliation Act of 2001.

Notwithstanding the foregoing, for purposes of Sections 2.26, 4.6 and 4.11, Compensation shall have the meaning set forth in Section 415(c)(3) of the Code, and shall include any payments made by the later of (a) 2-1/2 months after severance from employment or

(b) the end of the Limitation Year that includes the date of the severance from employment; provided that such payments are for services performed prior to such severance from employment and, absent a severance from employment, such payments would have been paid to the Participant while the Participant continued in employment with the Employer and such payments are regular compensation for services during the Participant's regular working hours, compensation for services outside the Participant's regular working hours (such as overtime or shift differential), commissions, bonuses or other similar compensation.

2.14. <u>Compensation Committee</u> means the Human Capital and Compensation Committee of the Board of Directors. For all purposes of the Plan, the Board acts through the Compensation Committee.

2.15. Effective Restatement Date means April 5, 2021.

2.16. <u>Eligible Associate</u> means an Associate who has attained the age of 21 and has completed at least one Year of Eligibility Service; provided, however, that a Full-Time Associate who has completed less than one Year of Eligibility Service shall be an Eligible Associate for purposes of Section 4.2 of the Plan (regarding Pre-Tax Savings Contributions and Roth Savings Contributions). Notwithstanding anything in the preceding sentence to the contrary, a Full-Time Associate who, before he or she has completed one Year of Eligibility Service, switches employment classifications and is no longer a Full-Time Associate shall remain an Eligible Associate for purposes of Section 4.2 of the Plan so long as he or she continues to be an Associate. Notwithstanding anything in this Section 2.16 to the contrary, an Associate shall be an Eligible Associate for purposes of Section 4.4 of the Plan (regarding Rollover Contributions).

2.17. <u>Eligibility Computation Period</u> means: (i) the initial Eligibility Computation Period of twelve consecutive months commencing on the date during a period of employment on which an Associate is first credited with an Hour of Service for the performance of duties for the Employer; and (ii) each and every full Plan Year during which the Associate is in the service of

the Employer, commencing with the Plan Year in which falls the last day of an Associate's initial Eligibility Computation Period.

2.18. Employer means the Plan Sponsor and all Affiliates.

2.19. <u>Employer Contribution Enrollment Date</u> means the first day of the first Enrollment Period beginning after an Associate has completed one Year of Eligibility Service.

2.20. <u>Employer Securities</u> means stock or other securities of L Brands or its former affiliates permitted to be held by the Plan under ERISA and the Code.

2.21. <u>Employer Securities Contribution Fund</u> means the Investment Fund consisting of Employer Securities and cash or cash equivalents needed to meet the obligations of such fund or for the purchase of common stock of L Brands.

2.22. <u>Enrollment and Change Designation</u> means an agreement, on a form or by a method prescribed by the Retirement Plan Committee, between a Participant and his or her Employer providing for reduction of the Participant's Compensation and the making of Pre-Tax Savings Contributions or Roth Savings Contributions by the Employer to the Plan and for designation of one or more Investment Funds.

2.23. Enrollment Period means each calendar month of a Plan Year.

2.24. ERISA means the Employee Retirement Income Security Act of 1974, as amended, and the regulations issued thereunder.

2.25. <u>Full-Time Associate</u> means an Associate who is regularly scheduled to work at least 30 hours per week and who is not a seasonal or otherwise temporary employee.

2.26. <u>Highly Compensated Associate</u> means any active Associate (or former associate, in accordance with Section 414(q)(6) of the Code) who (i) during the Plan Year or the preceding Plan Year was a 5% owner of the Employer, or (ii) during the preceding Plan Year received Compensation in excess of the limit in effect for such Plan Year under Section 414(q)(1) of the Code. An Associate shall be treated as a "5% owner" for any year if at any time during such year such Associate was a 5% owner (as defined in Section 416(i)(1) of the Code). The determination of who is a Highly Compensated Associate shall be made in accordance with Section 414(q) of the Code.

2.27. <u>Hour of Service</u> means: (i) each hour for which a person is paid or entitled to payment for the performance of duties for the Employer during the applicable computation period; (ii) each hour for which a person is paid or entitled to payment by the Employer on account of a period of time during which no duties are performed (irrespective of whether the employment relationship has terminated) due to vacation, holiday, illness, incapacity (including disability), jury or military duty or leave of absence; and (iii) each hour for which back pay, irrespective of mitigation of damages, is either awarded or agreed to by the Employer. For the purpose of determining Years of Eligibility Service, if a person's Hours of Service in an Eligibility Computation Period cannot be determined without undue administrative difficulty, such person shall be credited with 45 Hours of Service for each week in which he or she

completes one Hour of Service. Notwithstanding any provision in the Plan to the contrary, (i) not more than 501 Hours of Service shall be credited to any person on account of any single continuous period during which such person performs no duties; (ii) no credit shall be granted for any period with respect to which a person receives payment or is entitled to payment under a plan maintained solely for the purpose of complying with applicable worker's compensation or disability insurance laws; and (iii) no credit shall be granted for a payment which solely reimburses a person for medical or medically related expenses incurred by such person. Hours of Service shall be credited to the Plan Year in which payment for such Hours of Service is made. Determination and crediting of Hours of Service shall be under Department of Labor Regulations Sections 2530.200b-2 and 3.

2.28. Investment Funds means the funds described in Section 5.3.

2.29. Key Associate means a "key employee" as defined in Section 416(i) of the Code.

2.30. <u>Leased Employee</u> means any person (other than an Associate) who, pursuant to an agreement between the Employer and any other person ("leasing organization"), has performed services for the Employer (or for the Employer and related persons determined in accordance with Section 414(n) (6) of the Code) on a substantially full time basis for a period of at least one year, and such services are performed under the primary direction or control of the recipient. A person who is considered to be a "Leased Employee" of an Employer or related person shall not be considered an Associate for purposes of participating in the Plan or receiving any contribution or benefit under the Plan. However, if such person becomes a Participant as a result of subsequent employment with an Employer or related person, he or she shall receive vesting and eligibility service for his or her employment as a Leased Employee.

2.31. Limitation Year means the Plan Year.

2.32. <u>L Brands</u> means L Brands, Inc., a Delaware corporation, and any successor thereto.

2.33. <u>Matching Account</u> means the portion of the Account of a Participant consisting of Matching Contributions, as adjusted under the Plan, plus the Participant's stock bonus account as maintained under the Plan prior to July 1, 1990.

2.34. Matching Contributions means the amount contributed by the Employer under Section 4.3.

2.35. Non-Highly Compensated Associate means any Associate other than a Highly Compensated Associate.

2.36. <u>Non-Key Associate</u> means any Associate who is not a Key Associate.

2.37. Normal Retirement Date means the date a Participant attains age 65.

2.38. <u>Participant</u> means any person who has been admitted to participation in the Plan and has not ceased participation in the Plan.

2.39. <u>Plan</u> means the L Brands, Inc. 401(k) Savings and Retirement Plan, a profit sharing plan containing a cash or deferred arrangement, as set forth herein and as the same may from time to time be amended.

2.40. <u>Plan Sponsor</u> means L Brands Service Company, LLC, a Delaware limited liability company, and any successor thereto. Notwithstanding any provision herein to the contrary, any action permitted or required to be taken by the Plan Sponsor hereunder also may be taken by L Brands, through action of the Compensation Committee.

2.41. Plan Year means the calendar year.

2.42. <u>Post-Tax Savings Account</u> means the portion of the Account of a Participant consisting of Post-Tax Savings Contributions, as adjusted under the Plan.

2.43. <u>Post-Tax Savings Contributions</u> means the amount contributed by the Participant on an after-tax basis, as permitted by the Plan or other Affiliate's Plan, prior to January 1, 1989.

2.44. <u>Pre-Tax Savings Account</u> means the portion of the account of a Participant consisting of Pre-Tax Savings Contributions, as adjusted under the Plan.

2.45. <u>Pre-Tax Savings Contributions</u> means the amount contributed by the Employer as a result of a Participant's election on an Enrollment and Change Designation to reduce his or her Compensation, excluding amounts contributed as Roth Savings Contributions. Except as otherwise specifically provided in the Plan, the term "Pre-Tax Savings Contributions" when used herein shall include all Pre-Tax Savings Contributions that are treated as Catch-Up Contributions in accordance with Section 4.10.

2.46. <u>Retirement Account</u> means the portion of the Account of a Participant consisting of Retirement Contributions, as adjusted under the Plan.

2.47. <u>Retirement Contribution</u> means the amount contributed by the Employer under Section 4.1.

2.48. <u>Retirement Plan Committee</u> means the committee appointed by the Compensation Committee pursuant to Section 9.1.

2.49. <u>Rollover Account</u> means the portion of the Account of a Participant consisting of Rollover Contributions, as adjusted under the Plan.

2.50. <u>Rollover Contribution</u> means the amount contributed by an Associate as a rollover contribution in accordance with Section 402 of the Code, excluding Roth Rollover Contributions.

2.51. <u>Roth Conversion Amounts</u> means the amounts which (i) a Participant, (ii) a Beneficiary who is a surviving Spouse or (iii) an alternate payee who is a current or former Spouse has irrevocably elected to convert to Roth contributions, as provided in Section 4.13. All Roth Conversion Amounts shall be transferred directly to a Roth Conversion Subaccount within the Participant's (or Beneficiary's or alternate payee's) Roth Savings Account.

2.52. <u>Roth Conversion Subaccount</u> means the portion of a Roth Savings Account consisting of Roth Conversion Amounts, as adjusted under the Plan.

2.53. <u>Roth Rollover Account</u> means the portion of the Account of a Participant consisting of Roth Rollover Contributions, as adjusted under the Plan.

2.54. <u>Roth Rollover Contribution</u> means the amount contributed by an Associate as a Roth rollover contribution from a designated Roth account contained in an Eligible Retirement Plan (as defined in Section 8.1(b)(ii)) of another employer in accordance with Section 402A of the Code.

2.55. <u>Roth Savings Account</u> means the portion of the account of a Participant consisting of Roth Savings Contributions and Roth Conversion Amounts, as adjusted under the Plan.

2.56. <u>Roth Savings Contributions</u> means the amount that a Participant, as a result of the Participant's election on an Enrollment and Change Designation to reduce his or her Compensation, has irrevocably designated as being contributed in lieu of all or a portion of the Pre-Tax Savings Contributions that the Participant is otherwise eligible to make under the Plan, and that is treated by the Employer as includible in the Participant's gross income pursuant to Section 402A of the Code at the time the Participant would have received that amount in cash if the Participant had not elected to make the contribution. Except as otherwise specifically provided in the Plan, the term "Roth Savings Contributions" when used herein shall include all Roth Savings Contributions that are treated as Catch-Up Contributions in accordance with Section 4.10.

2.57. <u>Savings Contribution Enrollment Date</u> means the first day of the first Enrollment Period beginning after an Associate has become an Eligible Associate for purposes of Section 4.2 of the Plan (regarding Pre-Tax Savings Contributions and Roth Savings Contributions).

2.58. Self-Managed Brokerage Accounts means self-directed accounts described in Section 5.3.

2.59. Separation Date means the date on which an Associate incurs a severance from employment with the Employer for any reason.

2.60. <u>Spouse</u> means the individual to whom a Participant is legally married under the laws of the state or country in which the individual and the Participant were married, even if such marriage is not recognized under the laws of the state or country in which the Participant resides or is employed.

2.61. <u>Taxable Wage Base</u> means the contribution and benefit base in effect under Section 230 of the Social Security Act for the first day of any Plan Year.

2.62. <u>Top-Heavy Plan</u> means the Plan for any Plan Year if, on the determination date, the sum of the aggregate present value of the cumulative accrued benefits for Key Associates under all defined benefit plans in the Aggregation Group and the aggregate accounts of Key Associates under all defined contribution plans in the Aggregation Group exceeds 60% of a similar sum for all participants and beneficiaries (other than former Key Associates) under plans

in the Aggregation Group, determined in accordance with Section 416 of the Code. The ratio determined under this Section 2.63 shall: (a) take into account all distributions made during a 1-year period ending on the most recent determination date; provided that in the case of a distribution made for a reason other than severance from employment, death or disability, the phrase, "5-year period" shall be substituted for the phrase, "1-year period" and (b) shall not take into account any accrued benefit or account balance of an individual who has not performed services for the employer during a 1-year period ending on the determination date. The determination date for a Plan Year is the last day of the preceding Plan Year. For purposes of this Section 2.63, "Aggregation Group" means: (i) the Plan; (ii) any plan of an Affiliate in which a Key Associate (or any of a Key Associate's Beneficiaries) is a participant; (iii) any plan which enables any plan described in subsections (i) or (ii) to meet the requirements of Sections 401(a)(4) or 410 of the Code; (iv) any plan which was maintained by an Affiliate within the five years ending on the determination date and would but for the fact it was terminated be part of the Aggregation Group; and (v) any plan of any Affiliate designated by the Employer, the inclusion of which in the Aggregation Group would not cause the Aggregation Group to fail to meet the requirements of Sections 401(a)(4) and 410 of the Code.

2.63. <u>Total and Permanent Disability</u> means a physical or mental condition (i) of such severity and probable prolonged duration as to entitle the Participant to disability retirement benefits under the then-existing federal Social Security Act, or (ii) which qualifies as a total disability as defined under L Brands Long-Term Disability Plan for periods after the disability extension period. For purposes of the Plan, a Participant who is found to have incurred a Total and Permanent Disability shall be deemed to have incurred a Separation Date on the date his or her employment terminates in accordance with his or her Employer's employment policies. Notwithstanding the foregoing, a Participant will not be considered to have a Total and Permanent Disability under clause (ii) of the first sentence of this Section 2.64 until the date the Participant's employment terminates in accordance with his or her Employer's employment policies.

2.64. <u>Trust Agreement</u> means the trust agreement entered into between the Plan Sponsor and the Trustee to fund the benefits payable under the Plan, as the same presently exists and as it may from time to time hereafter be amended.

2.65. Trustee means the party or parties acting as such under the Trust Agreement.

2.66. <u>Trust Fund or Trust</u> means all of the assets of the Plan held by the Trustee at any time under the Trust Agreement.

2.67. <u>Valuation Date</u> means any day that the New York Stock Exchange is open for business or any other date chosen by the Retirement Plan Committee.

2.68. <u>Year of Eligibility Service</u> means an Eligibility Computation Period in which a person has completed 1,000 or more Hours of Service. For purposes of determining whether an Associate has completed a Year of Eligibility Service, an Associate who is on unpaid leave of absence taken under the Family and Medical Leave Act of 1993, or who is classified by the Employer as furloughed without pay during the period beginning April 5, 2020 and ending December 31, 2021, shall be credited with the Hours of Service with which a similarly-situated

Participant would normally have been credited if such leave or furlough had not occurred, as determined in accordance with reasonable procedures adopted from time to time by the Retirement Plan Committee.

2.69. <u>Year of Vesting Service</u> means a Plan Year during which a person (i) hired prior to January 1, 2010 is credited with at least 500 Hours of Service or (ii) hired on or after January 1, 2010, is credited with at least 1,000 Hours of Service. Notwithstanding anything in the preceding sentence to the contrary, a rehired Associate whose original date of hire is prior to January 1, 2010 and who has a rehire date of January 1, 2010 or later, shall be subject to clause (i) in the immediately preceding sentence. If an Associate has, prior to becoming an Associate, been employed by an Affiliate in a category of employment eligible to participate in an Affiliate Plan, then such Participant's service for Plan Years ending prior to the date such person becomes an Associate shall be determined under the provisions of such Affiliate Plan. For purposes of determining whether an Associate has completed a Year of Vesting Service, an Associate who is on unpaid leave of absence taken under the Family and Medical Leave Act of 1993, or who is classified by the Employer as furloughed without pay during the period beginning April 5, 2020 and ending December 31, 2021, shall be credited with the Hours of Service with which a similarly-situated Participant would normally have been credited if such leave or furlough had not occurred, as determined in accordance with reasonable procedures adopted from time to time by the Retirement Plan Committee.

ARTICLE III PARTICIPATION

3.1. Eligibility and Participation.

(a) An Eligible Associate shall become a Participant on the date which is the earliest of the date on which such person first (i) makes Pre-Tax Savings Contributions or Roth Savings Contributions to the Plan, as provided in Section 4.2, (ii) makes Rollover Contributions or Roth Rollover Contributions to the Plan, as provided in Section 4.4, or (iii) receives Retirement Contributions, as provided in Section 4.1.

(b) A Participant who ceases to be an Eligible Associate shall no longer be eligible to make Pre-Tax Savings Contributions, Roth Savings Contributions, Rollover Contributions and Roth Rollover Contributions, and to receive Retirement Contributions, Matching Contributions, allocations of forfeitures or any other Employer contribution as of the individual's Separation Date (or, if applicable, the date the Associate ceases to be an Eligible Associate). If a Participant incurs a Separation Date and is subsequently reemployed as an Eligible Associate and such individual either has incurred fewer than five consecutive Breaks in Service or has a vested interest in the Plan, then such Participant shall resume eligibility to make Pre-Tax Savings Contributions, Roth Savings Contributions, Rollover Contributions and Roth Rollover Contributions, and to receive Retirement Contributions, Matching Contributions, allocations of forfeitures or any other Employer contribution, as follows:

(i) If a rehired Eligible Associate was previously only a Participant with respect to his or her Rollover Account and/or Roth Rollover Account (i.e., prior to the

Participant's Separation Date, such Associate was not an Eligible Associate for purposes of Sections 4.1, 4.2 or 4.3 of the Plan), such rehired Eligible Associate shall become a Participant in the Plan in accordance with the terms of Subsection 3.1(c) of the Plan.

(ii) For purposes of Section 4.2 of the Plan (regarding Pre-Tax Savings Contributions and Roth Savings Contributions), if a rehired Eligible Associate previously was eligible to participate for purposes of Section 4.2, such individual shall be eligible to make Pre-Tax Savings Contributions and/or Roth Saving Contributions to the Plan as of the first day of the first Enrollment Period beginning after such individual's rehire date.

(iii) For purposes of Sections 4.1 and 4.3 of the Plan (regarding Retirement Contributions and Matching Contributions), a rehired Eligible Associate shall be eligible to receive Retirement Contributions and Matching Contributions as of the first day of the first Enrollment Period beginning after such individual's rehire date, provided however, if such individual had not previously completed one Year of Eligibility Service, such individual must first complete one Year of Eligibility Service and shall then be eligible as of such individual's Employer Contribution Enrollment Date.

(c) An Associate who was not a Participant in the Plan (or who participated only for purposes of Section 4.4 regarding Rollover Contributions and Roth Rollover Contributions) who incurs a Separation Date and is subsequently reemployed as an Associate shall be eligible to participate in the Plan as follows:

(i) A rehired Associate may make Rollover Contributions and Roth Rollover Contributions in accordance with Section 4.4 immediately.

(ii) An Associate who is rehired as a Full-Time Associate (A) shall be eligible to make Pre-Tax Savings Contributions and/or Roth Savings Contributions to the Plan as of the first day of the first Enrollment Period beginning after such individual's rehire date and (B) shall be eligible to receive Retirement Contributions and Matching Contributions as of the first day of the first Enrollment Period beginning after such individual's rehire date, provided however, if such individual had not previously completed one Year of Eligibility Service, such individual must first complete one Year of Eligibility Service and shall then be eligible to receive Retirement Contributions and Matching Contributions as of such individual's Employer Contribution Enrollment Date. For purposes of Subparagraph 3.1(c)(ii)(B), if the Associate is rehired prior to incurring one or more Breaks in Service, such Associate shall, for purposes of satisfying the one Year of Eligibility Service requirement to receive Retirement Contributions and Matching Contributions, resume accumulating Hours of Service upon such individual's date of reemployment.

(iii) An Associate who is not a Full-Time Associate who (A) has not earned at least one Year of Eligibility Service prior to his or her Separation Date and (B) is rehired prior to incurring one or more Breaks in Service shall, for purposes of satisfying the one Year of Eligibility Service requirement to make Pre-Tax Savings Contributions and/or Roth Savings Contributions and receive Retirement Contributions and Matching

Contributions, resume accumulating Hours of Service upon such individual's date of reemployment.

(iv) An Associate who is not a Full-Time Associate who (A) did not earn at least one Year of Eligibility Service prior to his or her Separation Date and (B) has incurred at least one Break in Service, but fewer than five consecutive Breaks in Service, shall be required to complete one Year of Eligibility Service upon his or her reemployment in order to participate in the Plan for purposes of Sections 4.1, 4.2 and 4.3.

(v) An Associate who is not a Full-Time Associate who (A) has earned at least one Year of Eligibility Service prior to his or her Separation Date, (B) has incurred fewer than 5 consecutive Breaks in Service and (C) is rehired prior to the first anniversary of such individual's original date of hire shall be eligible to participate in the Plan as of the first day of the first Enrollment Period beginning after the first anniversary of such individual's original date of hire.

(vi) Any Associate who (A) has earned at least one Year of Eligibility Service prior to his or her Separation Date, (B) has incurred fewer than 5 consecutive Breaks in Service and (C) is rehired after the first anniversary of his or her original hire date shall be eligible to participate in the Plan on such individual's date of reemployment and may make Pre-Tax Savings Contributions and/or Roth Savings Contributions as of the first Enrollment Period following such date.

(d) If an Associate (including a Participant) who incurs a Separation Date also incurs five or more consecutive Breaks in Service and was not vested in any portion of his or her Account or did not become a Participant in the Plan prior to the Associate's Separation Date, the Associate shall, upon reemployment, be required to satisfy the requirements of Section 3.1 as though such Associate had not previously been an Associate (i.e., without regard to any previously earned Years of Eligibility Service).

(e) If any Years of Eligibility Service are not required to be taken into account because of a period of Breaks in Service to which Sections 3.1(b), (c) or (d) applied, such Years of Eligibility Service shall not be taken into account in applying this Section 3.1 to any subsequent Breaks in Service.

3.2. <u>Waiver of Participation</u>. An Eligible Associate shall not have the right to waive participation unless the Retirement Plan Committee, in its sole discretion, determines to allow written waivers of participation. If such waivers are permitted, they shall be permitted on a nondiscriminatory basis and shall be effective on a year-to-year basis only. The Retirement Plan Committee retains the right not to permit waivers in any year or years, even if such waivers have been permitted in previous years.

3.3. <u>Change in Status</u>. In the event that a person who has been in the employ of an Affiliate in a category of employment not eligible for participation in the Plan subsequently becomes an Associate by reason of a change in status to a category of employment eligible for participation, he or she shall become a Participant as of the date on which his or her change in status occurs, if, on such date, he or she has otherwise satisfied the requirements for participation

in the Plan. If, on such date, he or she has not satisfied such requirements, he or she will become a Participant on the date of satisfaction of said requirements.

3.4. <u>Omission of Eligible Associate</u>. If, in any Plan Year, any Eligible Associate who should have been included as a Participant in the Plan is erroneously omitted and discovery of such omission is not made until after a contribution by the Employer for the Plan Year has been made and allocated, the Employer shall make a contribution with respect to the omitted Eligible Associate equal to the amount which the Eligible Associate would have received as an allocation of the Retirement Contribution had the Eligible Associate not been omitted, plus earnings thereon as determined in accordance with applicable law.

3.5. Inclusion of Ineligible Associate. If, in any Plan Year, any person who should not have been included as a Participant in the Plan is erroneously included and discovery of such incorrect inclusion is not made until after a Matching Contribution or Retirement Contribution for the Plan Year has been made and allocated, the Employer shall not be entitled to recover the Matching Contribution or Retirement Contribution made with respect to the ineligible person, and any earnings thereon, unless no deduction is allowable with respect to such Matching Contribution or Retirement Contribution. The amount contributed, together with any earnings thereon, with respect to the ineligible person shall constitute a forfeiture for the Plan Year in which the discovery is made.

3.6. <u>Cessation of Participation by ADS Transferred Associates</u>. Notwithstanding any provision of the Plan to the contrary, the participation of each ADS Transferred Associate shall cease as of his or her ADS Transfer Date.

ARTICLE IV CONTRIBUTIONS

4.1. Retirement Contributions.

(a) For each Plan Year, the Employer shall make non-service related Retirement Contributions to the Trust Fund, in cash and/or Employer Securities, on behalf of each Eligible Associate who (i) has completed at least one Year of Eligibility Service, (ii) completes 1,000 Hours of Service during such Plan Year, and (iii) is employed on the last day of such Plan Year, in an amount equal to (y) the sum of (A) 3% of such Eligible Associate's Compensation not in excess of the Taxable Wage Base and (B) 6% of such Eligible Associate's Compensation in excess of the Taxable Wage Base, or (z) such other greater or lesser amount as the Employer, in its absolute discretion, shall determine prior to the date on which the contributions are required to be made. The Plan Sponsor will notify the Trustee, in writing, of the amount of non-service related Retirement Contributions for each Plan Year.

(b) For each Plan Year, the Employer shall make service related Retirement Contributions to the Trust Fund, in cash and/or Employer Securities, on behalf of each Eligible Associate who (i) has completed at least one Year of Eligibility Service, (ii) completes 1,000 Hours of Service during such Plan Year, (iii) is employed on the last day of such Plan Year and (iv) has completed five or more Years of Vesting Service as of the last day of such Plan Year, in

an amount equal to (y) the sum of (A) 1% of such Eligible Associate's Compensation not in excess of the Taxable Wage Base and (B) 2% of Compensation in excess of the Taxable Wage Base, or (z) such greater or lesser amount as the Employer, in its absolute discretion, shall determine prior to the date on which the contributions are required to be made. The Plan Sponsor will notify the Trustee, in writing, of the amount of service related Retirement Contributions for each Plan Year.

(c) For purposes of determining whether an Associate is eligible for an allocation of Retirement Contributions pursuant to Subsections 4.1(a) or 4.1(b), an Associate who is on unpaid leave taken under the Family and Medical Leave Act of 1993, or who is classified by the Employer as furloughed without pay during the period beginning April 5, 2020 and ending December 31, 2021, shall be credited with the Hours of Service with which a similarly-situated Participant would normally have been credited if such leave or furlough had not occurred, as determined in accordance with reasonable procedures adopted from time to time by the Retirement Plan Committee, and any Associate on such leave or furlough on the last day of a Plan Year shall be treated as employed on such date.

(d) Notwithstanding any provision of the Plan to the contrary, for purposes of this Section 4.1, an Eligible Associate's Compensation shall not include any amounts earned prior to the Eligible Associate's Employer Contribution Enrollment Date (or, in the case of a rehired Associate, the applicable entry date provided for under Subsections 3.1(b) and (c) of the Plan).

(e) Notwithstanding any provision of the Plan to the contrary, an ADS Transferred Associate whose ADS Transfer Date occurs on or before December 31, 2012, shall be entitled to receive non-service related Retirement Contributions and service related Retirement Contributions for the Plan Year ending December 31, 2012, in the amounts set forth in Subsections (a) and (b) of this Section 4.1, regardless of whether such ADS Transferred Associate has completed the eligibility requirements for Retirement Contributions as set forth in Subsections (a)(iii) and (b)(iii) of this Section 4.1; provided, however, that the amount of such Retirement Contributions shall be based solely on Compensation received by such ADS Transferred Associate on or before the last day of the month in which his or her ADS Transfer Date occurs.

(f) Notwithstanding any provision of the Plan to the contrary, an Eligible Associate who is involuntarily terminated in connection with a reduction in force implemented by the Employer on or about April 1, 2013, and whose last day of employment with the Employer is on or before December 31, 2013, shall be entitled to receive non-service related Retirement Contributions and service related Retirement Contributions for the Plan Year ending December 31, 2013, in the amounts set forth in Subsections (a) and (b) of this Section 4.1, regardless of whether such Eligible Associate has completed the eligibility requirements for Retirement Contributions as set forth in Subsections (a)(iii) and (b)(iii) of this Section 4.1; provided, however, that the amount of such Retirement Contributions shall be based solely on Compensation received by such Eligible Associate on or before the last day of the month in which his or her involuntary termination occurs.

(g) Notwithstanding any provision of the Plan to the contrary, an Eligible Associate who is involuntarily terminated in connection with a reduction in force implemented by the

Employer on or about April 1, 2014, and whose last day of employment with the Employer is before December 31, 2014, shall be entitled to receive nonservice related Retirement Contributions and service related Retirement Contributions for the Plan Year ending December 31, 2014, in the amounts set forth in Subsections (a) and (b) of this Section 4.1, regardless of whether such Eligible Associate has completed the eligibility requirements for Retirement Contributions as set forth in Subsections (a)(ii) and (iii) and (b)(ii) and (iii) of this Section 4.1; provided, however, that the amount of such Retirement Contributions shall be based solely on Compensation received by such Eligible Associate on or before the last day of the month following the month in which his or her involuntary termination occurs.

(h) Notwithstanding any provision of the Plan to the contrary, an Eligible Associate who (i) is involuntarily terminated Without Cause during a Plan Year, or (ii) who is a 55/7 Retiree during a Plan Year, shall be entitled to receive non-service related Retirement Contributions and service related Retirement Contributions for such Plan Year in the amount set forth in Subsections (a) and (b) of this Section 4.1, regardless of whether such Eligible Associate has completed the eligibility requirements for Retirement Contributions as set forth in Subsections (a)(ii) and (iii) and (b)(ii) and (iii) of this Section 4.1 for such Plan Year.

For purposes of this Section 4.1(h), an involuntary termination "Without Cause" shall include (1) the Associate's death or Total and Permanent Disability, (2) termination resulting from staff reductions, a corporate or business restructuring, or a facility closing, or (3) an involuntary termination for any other reason which is not deemed to be a termination for cause under the Employer's personnel practices or policies.

For purposes of this Section 4.1(h), a "55/7 Retiree" shall mean an Eligible Associate who voluntarily terminates employment with the Employer, and, who, as of the last day of the Plan Year during which the Separation Date occurs, is at least age 55 and has completed at least seven (7) Years of Vesting Service.

4.2. Pre-Tax and Roth Savings Contributions.

(a) An Associate shall be eligible to make Pre-Tax Savings Contributions and/or Roth Savings Contributions as of the Associate's Savings Contribution Enrollment Date. A Participant who is no longer a Full-Time Associate but remains an Associate shall continue to be eligible to make Pre-Tax Savings Contributions and/or Roth Savings Contributions. A rehired Associate shall be eligible to make Pre-Tax Savings Contributions and/or Roth Savings Contributions in accordance with the terms of Subsections 3.1(b) and (c).

(b) A Participant may change the amount of his or her Pre-Tax Savings Contributions and/or Roth Savings Contributions, or discontinue Pre-Tax Savings Contributions and/or Roth Savings Contributions, at any time by submitting a new Enrollment and Change Designation. The Enrollment and Change Designation shall become effective subsequent to such election when administratively practicable and shall provide for reduction of the Compensation of such Participant and a corresponding contribution to the Plan by the Employer as a Pre-Tax Savings Contribution, which shall be allocated to the Participant's Pre-Tax Savings Account, and/or a Roth Savings Contribution, which shall be allocated to the Participant shall be entitled to direct in his or her Enrollment and Change Designation that

an aggregate contribution (including any Pre-Tax Savings Contributions and Roth Savings Contributions) of up to 75% of the Participant's Compensation be made during each payroll period of the Employer. Notwithstanding any provision of the Plan to the contrary, for purposes of this Subsection 4.2(b), a Participant's Compensation shall not include any amounts earned prior to the Participant's Savings Contribution Enrollment Date (or, in the case of a rehired Associate, the applicable entry date provided for under Subsections 3.1(b) and (c) of the Plan). In the event that a Participant does not designate on his Enrollment and Change Designation whether the contributions elected to be made are Pre-Tax Savings Contributions or Roth Savings Contributions, all contributions elected on such application shall be deemed for all purposes of the Plan to be Pre-Tax Savings Contributions.

(c) A Participant's Pre-Tax Savings Contribution and Roth Savings Contributions for a Plan Year, when aggregated with any pre-tax and Roth savings contributions made to any plan maintained by an Affiliate, shall not exceed the dollar limit established under Section 402(g) of the Code (except to the extent permitted under the Catch-Up Contribution provisions set forth in Section 4.10 of the Plan and Section 414(v) of the Code). Any Pre-Tax Savings Contribution or Roth Savings Contribution which would exceed any applicable limit shall be returned to the Participant, together with earnings thereon through the end of the calendar year in which such contribution was made, by April 15th of the next following Plan Year; provided that, if such excess contributions were the result of contributions made both to the Plan and to a plan or plans not maintained by the Employer or an Affiliate, the Participant must first notify the Retirement Plan Committee of the amount of such excess allocable to the Plan by March 1 of the year following the year in which the excess contribution occurred. To the extent that any Pre-Tax Savings Contributions or Roth Savings Contributions are distributed to a Highly Compensated Associate pursuant to this Subsection 4.2(c), any Matching Contributions allocated to the Highly Compensated Associate with respect to such distributed Pre-Tax Savings Contributions and Roth Savings Contributions were made on behalf of a Participant for the Plan Year to which the excess contributions and Roth Savings Contributions were made on behalf of a Participant for the Plan Year to which the excess contributions and Roth Savings Contributions were made on behalf of a Participant for the Plan Year to which the excess contributions relate, any excess Pre-Tax Savings Contributions will be returned to the Participant first, subject to any alternative procedure as may be adopted by the Retirement Plan Committee from time to time.

4.3. Matching Contributions.

(a) The Employer shall make a Matching Contribution in cash and/or Employer Securities for each Participant who (i) makes a Pre-Tax Savings Contribution or Roth Savings Contribution and (ii) has completed at least one Year of Eligibility Service. The Matching Contribution shall equal 100% of the Participant's Pre-Tax Savings Contributions and Roth Savings Contributions for a Plan Year (not exceeding 4% of the Participant's Compensation) made on or after the Participant's Employer Contribution Enrollment Date (or, in the case of a rehired Associate, the applicable entry date provided for under Subsections 3.1(b) and (c) of the Plan). Matching Contributions shall be allocated to the Matching Accounts of the Participants with respect to whom the contributions are made.

(b) Except as otherwise provided in Section 4.12, the Plan is intended to satisfy the safe harbor method of meeting the nondiscrimination requirements set forth under Section 401(k)(12) and Section 401(m)(11) of the Code. Accordingly, the Employer shall, within a

reasonable period before any year, give each Eligible Associate written notice of his or her rights and obligations under this Section 4.3 which notice is sufficiently accurate and comprehensive to apprise the Eligible Associate of such rights and obligations, and written in a manner calculated to be understood by the average employee eligible to participate.

(c) Notwithstanding any provision of the Plan to the contrary, for purposes of this Section 4.3, (i) any Compensation earned by a Participant for any period during which the Participant is not eligible to receive Matching Contributions shall not be taken into account for purposes of determining the amount of the Participant's Matching Contribution, and (ii) any Compensation earned by a Participant for periods during which the Participant is eligible to receive Matching Contributions or Roth Savings Contributions were made by the Participant shall be taken into account for purposes of determining the amount of the Participant's Matching Contributions or Roth Savings Contributions were made by the Participant shall be taken into account for purposes of determining the amount of the Participant's Matching Contribution.

4.4. Rollover Contributions.

(a) The Retirement Plan Committee may authorize the Trustee to accept a Rollover Contribution directly from an Associate or from an Eligible Retirement Plan (as defined in Paragraph 8.1(b)(ii)) on behalf of the Associate, provided that such Rollover Contribution is an Eligible Rollover Distribution (as defined in Paragraph 8.1(b)(i)). A Rollover Contribution received directly from an Associate must be paid to the Trustee in cash within 60 days after the date the Associate received the amounts representing the Rollover Contribution from an Eligible Retirement Plan. A Rollover Contribution received directly from another Eligible Retirement Plan may be paid to the Trustee in cash or kind, provided that the Trustee may refuse to receive any in kind transfer if the Trustee finds the in kind assets unacceptable. Rollover Contributions described in this Section 4.4(a) shall be posted to the applicable Associate's Rollover Account as soon as administratively feasible after the date the contribution is received by the Trustee. A Rollover Contribution may include a rollover from an Associate's after-tax employee contribution portion of a qualified plan described in Section 401(a) or 403(a) of the Code to the Plan, provided that such distribution is eligible for a tax free rollover within the meaning of Section 402(c)(2) of the Code and is made by means of a direct trustee-to-trustee transfer. The Associate shall present a written certification to the Trustee that the Associate's Rollover Contribution satisfies the requirements of this Subsection 4.4(a).

(b) The Retirement Plan Committee may authorize the Trustee to accept a Roth Rollover Contribution directly from an Associate or from a designated Roth account (as described in Section 402A of the Code) of an Eligible Retirement Plan (as defined in Paragraph 8.1(b)(ii)) on behalf of the Associate, provided that such Roth Rollover Contribution is an Eligible Rollover Distribution (as defined in Paragraph 8.1(b)(i)). A Roth Rollover Contribution received directly from an Associate must be paid to the Trustee in cash within 60 days after the date the Associate received the amounts representing the Roth Rollover Contribution from an Eligible Retirement Plan. A Roth Rollover Contribution received directly from another Eligible Retirement Plan may be paid to the Trustee in cash or kind, provided that the Trustee may refuse to receive any in kind transfer if the Trustee finds the in kind assets unacceptable. Roth Rollover Contributions described in this Section 4.4(b) shall be posted to the applicable Associate's Roth Rollover Account as soon as administratively feasible after the date the contribution is received

by the Trustee. The Associate shall present a written certification to the Trustee that the Associate's Roth Rollover Contribution satisfies the requirements of this Subsection 4.4(b).

(c) Rollover Contributions shall be accounted for separately and shall be credited to an Associate's Rollover Account. Furthermore, within an Associate's Rollover Account, any Rollover Contributions which are rolled over as an after-tax employee contribution to the Plan shall be accounted for separately. Roth Rollover Contributions shall be accounted for separately and shall be credited to an Associate's Roth Rollover Account.

(d) An Associate who has made a Rollover Contribution or Roth Rollover Contribution shall be deemed to be a Participant with respect to his or her Rollover Account or Roth Rollover Account, respectively, even if he or she is not otherwise a Participant.

4.5. Timing of Contributions.

All Pre-Tax Savings Contributions and Roth Savings Contributions shall be made no later than the earlier of (i) the earliest date after the reduction of Associates' Compensation on which the Pre-Tax Savings Contributions and Roth Savings Contributions can reasonably be segregated from the Employer's general assets or (ii) the 15th business day of the month following the month of the date of payroll deduction on which such Associates' Compensation was reduced. Retirement Contributions and Matching Contributions shall be made no later than the due date (including extensions) of the Federal income tax return of the Plan Sponsor for the fiscal year of the Plan Sponsor which includes the last day of the Plan Year for which such contribution is made. All contributions shall be paid over to the Trustee and shall be invested by the Trustee in accordance with the Plan and the Trust Agreement.

4.6. Annual Additions and Limitations.

(a) Notwithstanding any other provisions of the Plan, in no event shall the annual additions to a Participant's Account for any Plan Year exceed the lesser of (i) \$58,000 as adjusted in accordance with Section 415(d) of the Code or (ii) 100% of such Participant's Compensation. All amounts contributed to any defined contribution plan maintained by any Affiliate other than a plan described in Section 415(c)(6) of the Code shall be aggregated with contributions made by the Employer under the Plan in computing any Participant's annual additions for a Plan Year. In no event shall the amount allocated to the Account of any Participant be greater than the maximum amount allowed under Section 415 of the Code with respect to any combination of plans without disqualification of any such plan.

(b) For purposes of this Section 4.6, the term "annual additions" means the sum for any Plan Year of all Employer and Associate contributions and forfeitures allocated to the Account of a Participant, and amounts derived from contributions paid or accrued that are attributable to post-retirement medical benefits allocated to the separate account of a Participant who is a key employee, as defined in Section 419A(d)(3) of the Code, under a welfare benefit fund, as defined in Section 419(e) of the Code, maintained by the Plan Sponsor or an Affiliate.

(c) If a Participant's annual additions would otherwise exceed the limitations set forth in this Section 4.6, such excess shall be corrected in accordance with Revenue Procedure

2008-50 or such other guidance as may be prescribed by the Commissioner of the Internal Revenue Service.

4.7. <u>Fail-Safe Allocations of Retirement Contributions</u>. Notwithstanding anything in the Plan to the contrary, if the Plan would otherwise fail to meet the requirements of Sections 401(a)(4) or 410(b) of the Code because service related or non-service related Retirement Contributions have not been allocated to a sufficient number or percentage of Participants for a Plan Year, then the following rules shall apply:

(a) The group of Participants eligible to share in the service related Retirement Contributions for the Plan Year shall be expanded to include the minimum number of Participants (who have less than five Years of Vesting Service and so would not otherwise be eligible to share in the service-related Retirement Contributions) as are necessary to satisfy the applicable test. The specific Participants who shall become eligible under the terms of this Subsection 4.7(a) shall be those Participants who are actively employed on the last day of the Plan Year, have completed 1,000 Hours of Service during the Plan Year, who are Non-Highly Compensated Associates with at least four Years of Vesting Service, and who, when compared to similarly situated Participants, have the earliest original hire date.

(b) The group of Participants eligible to share in the non-service related Retirement Contribution for the Plan Year shall be expanded to include the minimum number of Participants (who failed to complete 1,000 Hours of Service during the Plan Year and so would not otherwise be eligible to share in the Retirement Contribution) as are necessary to satisfy the applicable test. The specific Participants who shall become eligible under the terms of this Subsection 4.7(b) shall be those who are actively employed on the last day of the Plan Year and who, when compared to similarly situated Participants, have completed the greatest number of Hours of Service in the Plan Year.

(c) If after application of the previous Subsection 4.7(b), the applicable test is still not satisfied, then the group of Participants eligible to share in the non-service related Retirement contribution for the Plan Year shall be further expanded to include the minimum number of former Participants (who are not employed on the last day of the Plan Year and so would not otherwise be eligible to share in the Retirement Contribution) as are necessary to satisfy the applicable test. The specific former Participants who shall become eligible under the terms of this Subsection 4.7(c) shall be those former Participants who are Non-Highly Compensated Associates with at least three Years of Vesting Service who, when compared to similarly situated former Participants, have completed the greatest number of Hours of Service in the Plan Year before terminating employment.

(d) Nothing in this Section 4.7 shall permit the reduction of a Participant's benefit. Therefore, any amounts that have previously been allocated to Participants may not be reallocated to satisfy these requirements. In the event additional allocations are required, the Employer shall make an additional contribution equal to the additional allocations, even if such additional amount exceeds the amount which would be deductible under Section 404 of the Code. Any adjustment to the allocations pursuant to this Section 4.7 shall be made by the October 15 of the Plan Year after the Plan Year to which the allocations relate and shall be considered to be made as of the last day of the Plan Year to which the allocations relate.

4.8. <u>Forfeitures</u>. Forfeitures (other than Forfeitures reinstated pursuant to Section 7.2) shall be applied to reduce contributions or to pay Plan expenses in such manner as the Retirement Plan Committee determines.

4.9. Exclusive Benefit; Refund of Contributions.

(a) All contributions made by the Employer are made for the exclusive benefit of the Participants and their Beneficiaries, and except as otherwise permitted by law, such contributions shall not be used for or diverted to purposes other than for the exclusive benefit of the Participants and their Beneficiaries, including the costs of maintaining and administering the Plan and Trust Fund.

(b) Notwithstanding the foregoing, amounts contributed to the Trust Fund by the Employer may be refunded to the Employer under the following circumstances and subject to the following limitations:

(i) To the extent that a Federal income tax deduction is disallowed for any contribution made by the Employer, the Trustee shall return to the Employer the amount so disallowed within one year of the date of such disallowance; and

(ii) In the event a contribution is made, in whole or in part, by reason of a mistake of fact, the Trustee shall return to the Employer so much of such contribution as is attributable to the mistake of fact within one year after the payment of the contribution to which the mistake applies.

(c) In the case of a refund described in Paragraph (i) or (ii) of Subsection 4.9(b), the amount to be returned shall be the amount contributed over the amount that would have been contributed had there not occurred a mistake of fact or a mistake in determining the deduction. Earnings attributable to the excess contribution may not be returned to the Employer, but losses attributable thereto must reduce the amount to be so returned. Furthermore, if the withdrawal attributable to the mistaken contribution would cause the balance of the individual account of any Participant to be less than the balance which would have been in the account had the mistaken amount not been contributed, then the amount to be returned to the Employer must be limited to avoid such reduction.

4.10. <u>Catch-Up Contribution</u>. All Participants who have elected to make Pre-Tax Savings Contributions and/or Roth Savings Contributions for a Plan Year and who will attain age 50 on or before the last day of their taxable year are eligible to make a Catch-Up Contribution during such Plan Year in accordance with, and subject to the limitations of, Section 414(v) of the Code. In furtherance of, but without limiting, the foregoing, to the extent permitted under Section 414(v) of the Code, Pre-Tax Savings Contributions and/or Roth Savings Contributions for such a Participant for a Plan Year which exceed the maximum contribution permitted under Section 4.2(c) for the Plan Year or the maximum contribution permitted under Section 4.6 for the Plan Year shall be treated as a Catch-Up Contribution; provided, however, that whether Pre-Tax Savings Contributions and Roth Savings Contributions are in excess of any applicable limit and therefore shall be treated as a Catch-Up Contribution shall be determined as of the end of the Plan Year. The Catch-Up Contribution will not be taken into account for purposes of the

limitation on the maximum amount of the Participant's Pre-Tax Savings Contributions and Roth Savings Contributions for a Plan Year under Subsection 4.2(c) of the Plan (and Section 402(g) of the Code) or the limitation on contributions for a Plan Year under Section 4.6 of the Plan (and Section 415(c) of the Code). Further, by allowing such Catch-Up Contributions, the Plan shall not be treated as failing to satisfy the provisions of the Plan implementing the requirements of Section 401(k)(12), 410(b) or 416 of the Code, as applicable. Except as provided in the preceding two sentences, Catch-Up Contributions shall be treated in the same manner as Pre-Tax Savings Contributions and Roth Savings Contributions for all purposes under the Plan, including for purposes of Matching Contributions under Section 4.3 of the Plan.

4.11. <u>Minimum Top-Heavy Contributions</u>. For each Plan Year that the Plan is a Top-Heavy Plan, the Employer shall contribute, for and on behalf of each person who is a Participant on the last day of the Plan Year, not less than the lesser of (a) 3% of such Participant's Compensation (or 5% of such Participant's Compensation if the Participant also participates in a top-heavy defined benefit plan in the Aggregation Group (as defined in Section 2.63)); or (b) such Participant's Compensation multiplied by a fraction, determined with respect to the Key Associate for whom the fraction is greatest, the numerator of which is the contribution for such Key Associate for the Plan Year (including Pre-Tax Savings Contributions and Roth Savings Contributions for any Participant to any plan included in the Aggregation Group shall be taken into account, in accordance with Section 416 of the Code. Pre-Tax Savings Contributions and Roth Savings Contributions for any Participant other than a Participant who is a Key Associate shall not be taken into account in determining such 3% (or 5%) contribution.

4.12. Nondiscrimination Test for Disaggregated Pre-Tax and Roth Savings Contributions.

(i) Notwithstanding any provision of the Plan to the contrary, the portion of the Plan covering Associates who are eligible to make Pre-Tax Savings Contributions and Roth Savings Contributions under Section 4.2, but who are not yet eligible to receive Matching Contributions under Section 4.3, shall be disaggregated from the portion of the Plan otherwise intended to satisfy the safe harbor method of meeting the nondiscrimination requirements and shall satisfy the nondiscrimination test of Code Section 401(k) (described in this Section 4.12) for each Plan Year in accordance with Treasury Regulation Sections 1.401(k)-1(b)(4)(iv) and 1.401(k)-3(h)(3). The nondiscrimination test provisions described in this Section 4.12 shall be applied only with regard to the disaggregated portion of the Plan for each Plan Year.

(ii) In order to ensure that the non-discrimination test under this Section 4.12 is satisfied, the Retirement Plan Committee shall direct the Employer to adjust the Pre-Tax Savings Contributions and/or Roth Savings Contributions for the Plan Year with regard to the disaggregated portion of the Plan to the extent necessary to meet the requirements of Code Section 401(k) and shall instruct the Employer as to how such adjustment shall be made. An adjustment to Pre-Tax Savings Contributions and/or Roth Savings Contributions shall be accomplished by: (i) requiring Highly Compensated Associates to reduce (or eliminate) the Pre-Tax Savings Contributions and/or Roth

Savings Contributions to be made on his or her behalf for the Plan Year; (ii) returning Pre-Tax Savings Contributions and/or Roth Savings Contributions made on behalf of Highly Compensated Associates to such Highly Compensated Associates as of the end of the Plan Year; (iii) making a qualified nonelective contribution (as defined in Treasury Regulation Section 1.401(k)-6) for Participants who are not Highly Compensated Associates, and who elected to have Pre-Tax Savings Contributions and/or Roth Savings Contributions made for the Plan Year in accordance with Section 4.2 of the Plan; or (iv) taking such other actions as the Retirement Plan Committee deems appropriate.

(iii) The non-discrimination requirements of Code Section 401(k) require that, in each Plan Year, the disaggregated portion of the Plan must satisfy one of the following tests:

- (i) The Actual Deferral Percentage (defined below) of the Highly Compensated Associates for the Plan Year is not more than the Actual Deferral Percentage of all other eligible Associates for the Plan Year, multiplied by 1.25; or
- (ii) The excess of the Actual Deferral Percentage of the Highly Compensated Associates for the Plan Year over the Actual Deferral Percentage of the other eligible Associates for the Plan Year is not more than two (2) percentage points, and the Actual Deferral Percentage of the Highly Compensated Associates for the Plan Year is not more than the Actual Deferral Percentage of all other eligible Associates for the immediately preceding Plan Year, multiplied by two (2).

(iv) The Plan elects to utilize the current year testing method. Notwithstanding the foregoing, the Retirement Plan Committee may elect to use the immediately preceding Plan Year's Actual Deferral Percentage for Eligible Associates who are not Highly Compensated Associates, instead of their Actual Deferral Percentage for the current Plan Year, in applying the tests described above. Such election shall be made in accordance with Code Section 401(k)(3)(A) and applicable Treasury Regulations and guidance issued by the Internal Revenue Service.

(v) The "Actual Deferral Percentage" is the average of the ratios, calculated separately for each Associate in the disaggregated portion of the Plan in accordance with Treasury Regulation Section 1.401(k)-2(a)(3), of the amount of Pre-Tax Savings Contributions and Roth Savings Contributions that are credited under the Plan on behalf of the Eligible Associate for the Plan Year (including, with respect to Highly Compensated Associates, any excess deferrals described in Treasury Regulation Section 1.402(g)-1(a)), to the Associate's Compensation for the Plan Year.

(vi) Notwithstanding the foregoing, if the test described in this Section 4.12 is not satisfied for a Plan Year, the Retirement Plan Committee may use any other test permitted under Code Section 401(k) to determine whether the disaggregated portion of the Plan meets the nondiscrimination requirements of Code Section 401(k). At least annually, the Retirement Plan Committee shall evaluate the Plan's operation to assure

that Pre-Tax Savings Contributions and Roth Savings Contributions elected by Associates who are Highly Compensated Associates do not cause the limitations of this Section 4.12 to be exceeded. The Employer, in the Retirement Plan Committee's sole discretion, may make a qualified nonelective contribution to the Pre-Tax Savings Contribution Accounts of Associates who are not Highly Compensated Associates and who elected to have Pre-Tax Savings Contributions and/or Roth Savings Contributions made for the Plan Year, allocated among those Accounts in proportion to those Associates' relative Compensation to assist the disaggregated portion of the Plan in satisfying the limitations of Section 4.12. To the extent that Associates who are Highly Compensated Associates' Pre-Tax Savings Contribution and Roth Savings Contributions would, if carried out, cause the limitations of Section 4.12 to be exceeded, the elections shall be adjusted so that Pre-Tax Savings Contributions and Roth Savings Contributions (and Compensation allocable to those contributions) are returned to Associates who are Highly Compensated Associates until those contributions satisfy the limitations of Section 4.12. Refunds will be made in order of the actual dollar amount of Pre-Tax Savings Contributions and Roth Savings Contributions by Associates who are Highly Compensated Associates starting with those who have the highest contribution amounts, and reducing them to the next-highest contribution amounts (or if less, to the extent necessary to prevent violation of Code Section 401(k)(3)). This process shall be repeated until the limits of Section 4.12 are satisfied.

The portion of any Pre-Tax Savings Contribution or Roth Savings Contribution that has been reduced pursuant to the foregoing shall be regarded as an "Excess Savings Contribution". To the extent necessary to ensure compliance with the limitations of Section 4.12; the Trustee shall return all Excess Savings Contributions, together with all income allocable thereto through the end of the calendar year in which such contribution was made, to the Associate on whose behalf the contributions were made within one year after the end of the Plan Year for which the contributions were made in accordance with Section 4.12(viii). In the event both Pre-Tax Savings Contributions and Roth Savings Contributions were made on behalf of a Participant for the Plan Year to which the Excess Savings Contributions relate, any excess Pre-Tax Savings Contributions will be returned to the Participant first, subject to any alternative procedure as may be adopted by the Retirement Plan Committee from time to time.

(vii) The portion of this Plan intended to satisfy the safe harbor method of meeting the nondiscrimination requirements shall not be aggregated with any other plans that do not satisfy the safe harbor method of meeting the nondiscrimination requirements (regardless of whether any such plan is merged into this Plan pursuant to Section 12 of the Plan). If the Employer maintains more than one plan qualified under Code Section 401(a), and if the non-safe harbor plans are aggregated for purposes of satisfying the coverage or non-discrimination requirements of Code Section 401(a) or 410(b)(1)(A) or (B), all qualified cash or deferred arrangements contained in such non-safe harbor plans shall be aggregated for purposes of performing the non-discrimination test for Pre-Tax Savings Contributions and Roth Savings Contributions. If a Highly Compensated Associate is simultaneously eligible to participate in more than one non-safe harbor plan of the Employer, all Pre-Tax Savings Contributions and Roth Savings

Contributions made by the Highly Compensated Associate under all such non-safe harbor plans shall be aggregated for purposes of performing the test described in Section 4.12.

(viii) If Pre-Tax Savings Contributions and Roth Savings Contributions of Highly Compensated Associates are required to be reduced as a result of the non-discrimination test described in Section 4.12, the Excess Savings Contributions and income attributable to those contributions shall be distributed to the Highly Compensated Associates within 2-1/2 months after the close of the Plan Year to which the Pre-Tax Savings Contributions and/or Roth Savings Contributions relate. In determining the amount of the distributions required under this Section 4.12(viii), the Retirement Plan Committee shall use the leveling method described in Code Section 401(k)(8)(C) and applicable Treasury Regulations thereunder, or any other method allowed by the Internal Revenue Service. The amount of income attributable to excess contributions were allocated that bears the same ratio as the amount of excess contributions bears to the total balance of that Account. The distributions required under this Section 4.12(viii) may be made without the consent of the Associate or his or her Spouse and may be made without regard to any "qualified domestic relations order" as defined in Section 414(p) of the Code.

4.13. <u>Roth Conversion Amounts</u>. Notwithstanding anything in the Plan to the contrary, a Participant, a Beneficiary who is a surviving Spouse, or an alternate payee who is a current or former Spouse may make an election, at the time and in the manner prescribed by the Retirement Plan Committee, but in no event more than two times per calendar year, to roll over directly to a Roth Conversion Subaccount within his Roth Savings Account under the Plan all or any portion of his Pre-Tax Savings Account, Post-Tax Savings Account and/or Rollover Account (whether or not currently distributable under the Plan). Any election made pursuant to this Section 4.13 shall constitute an irrevocable election to convert the amounts to be rolled over to Roth contributions, and such Roth Conversion Amounts shall be treated by the Plan as includible in the electing individual's income at the time of conversion. Roth Conversion Amounts shall be reated by the Plan as includible in the same time and in the same order and classification as applied to such amounts prior to their conversion. Notwithstanding the foregoing, if all or a portion of an amount rolled over into a Participant's Roth Conversion Subaccount is later determined to be an excess contribution described in Section 4.2(c) or an Excess Savings Contribution described in Section 4.12, and the excess amount (plus applicable earnings) is to be distributed from the Plan, then the excess amount (plus applicable earnings) may be distributed from the Roth Conversion Subaccount, even if the amount was an otherwise nondistributable amount at the time of the rollover, if determined necessary by the Retirement Plan Committee and to the extent permitted by law.

ARTICLE V INVESTMENT OF THE TRUST FUND

5.1. <u>Investment Direction</u>. Each Participant shall have the right (subject to Section 5.4(c) and subject to the limitations on managed accounts described in this Article V) to direct that (i) future contributions to the Participant's Account be invested in one or more of the

Investment Funds, and (ii) the existing balance in the Participant's Account be invested in one or more Investment Funds. A Participant may make or change an investment direction by submitting an Enrollment and Change Designation at such times and by such method as determined by the Retirement Plan Committee or the person or entity designated by the Retirement Plan Committee to maintain Plan records. The provisions of this ARTICLE V pertaining to Participant-directed investments are intended to comply with the provisions of Section 404(c) of ERISA and the regulations thereunder.

5.2. <u>Absence of Investment Direction</u>. Any portion of a Participant's Account as to which the Participant fails to provide an investment direction, or as to which the Participant's investment direction shall not be given effect as a result of the restrictions set forth in Section 5.4(c), shall be invested by the Trustee in a fund which constitutes a "qualified default investment alternative" (as defined in Department of Labor Regulations Section 2550.404c-5(e)).

5.3. Investment Funds. One of the Investment Funds shall be the Employer Securities Contribution Fund. The Retirement Plan Committee shall direct the Trustee to create and maintain three or more additional Investment Funds according to investment criteria established by the Retirement Plan Committee. The Retirement Plan Committee shall have the right to direct the Trustee to merge or modify any existing Investment Funds, or to create additional Investment Funds, after notice to Participants investing in the affected Investment Funds. The Trust Fund shall be divided into (a) the Employer Securities Contribution Fund, (b) Investment Funds established by the Retirement Plan Committee, and (c) self-directed accounts ("Self-Managed Brokerage Accounts shall be treated as Investment Funds for all purposes under the Plan except that the investment authority with respect to each Self-Managed Brokerage Account shall belong to the Participant who established the Self-Managed Brokerage Account rather than the Retirement Plan Committee or any other fiduciary. No person who is otherwise a fiduciary or involved in the administration of the Plan shall have any responsibility to review any Self-Managed Brokerage Account. Whether a particular investment is within the category of one or another of such Investment Funds shall be determined by the Retirement Plan Committee solely in its discretion.

5.4. Investment in Employer Securities.

(a) One of the principal purposes of the Plan is to provide Participants with ownership interests in the Employer, and to the extent practicable, all available assets of the Employer Securities Contribution Fund shall be used to purchase Employer Securities, which shall be held by the Trustee until distribution or sale for distribution of cash to Participants or Beneficiaries or until disposition is required to implement changes in investment designations.

(b) Notwithstanding any provision of this ARTICLE V to the contrary, but subject to Section 5.4(c), each Participant shall have the right no less frequently than once per calendar quarter to make or change an investment direction with respect to the portion of the balance of the Participant's Pre-Tax Savings Account, Post-Tax Savings Account and Roth Savings Account that is invested in Employer Securities and each Participant with at least three Years of Vesting Service shall have the right no less frequently than once per calendar quarter to make or change an investment direction with respect to the portion of the balance of the Participant's Matching Account and Retirement Account that is invested in Employer Securities. In addition,

all or any portion of any other Investment Funds (other than the Self-Managed Brokerage Account) may consist of Employer Securities.

(c) Notwithstanding any provision of this ARTICLE V to the contrary, effective with respect to elections made on or after June 24, 2021, a Participant's election under clause (i) of Section 5.1 may not provide that more than twenty-five percent (25%) of the Participant's future contributions shall be invested in the Employer Securities Contribution Fund. To the extent that a Participant's election made on or after June 24, 2021 under clause (i) of Section 5.1 provides that more than twenty-five percent (25%) of the Participant's election made on or after June 24, 2021 under clause (i) of Section 5.1 provides that more than twenty-five percent (25%) of the Participant's future contributions shall be invested in the Employer Securities Contribution Fund, the contributions to the Plan in excess of the twenty-five percent (25%) limit shall be invested in accordance with Section 5.2. Effective as of June 24, 2021, a Participant's request for a fund transfer under clause (ii) of Section 5.1 shall be given effect only if less than twenty-five percent (25%) of the value of such Participant's total Account is currently allocated to the Employer Securities Contribution Fund and only to the extent that giving effect to the request will not cause more than twenty-five percent (25%) of the value of such Participant's total Account.

5.5. <u>Voting Employer Securities</u>. The Retirement Plan Committee shall have the power to direct the Trustee in the voting of all Employer Securities held by the Trustee. All voting of Employer Securities shall be in compliance with all applicable rules and regulations of the Securities and Exchange Commission and all applicable rules of or any agreement with any stock exchange on which the Employer Securities being voted are traded. Notwithstanding the foregoing, the Retirement Plan Committee may, in its sole discretion and at any time or from time to time, permit Participants and Beneficiaries to direct the manner in which all or the vested portion of any Employer Securities allocated to their Accounts shall be voted on such matters as the Retirement Plan Committee permits. Upon timely receipt of directions under this Section 5.5 from the Retirement Plan Committee, Participant or Beneficiary, the Trustee shall vote all Employer Securities as directed. If the Trustee does not receive timely directions from the Retirement Plan Committee, Participant or Beneficiary under this Section 5.5, the Trustee shall not vote the Employer Securities with respect to which direction was not given.

5.6. <u>Tender Offers</u>. Each Participant and Beneficiary shall have the sole right to direct the Trustee as to the manner in which to respond to a tender or exchange offer for Employer Securities allocated to such person's Account. The Retirement Plan Committee shall use its best efforts to notify or cause to be notified each Participant and Beneficiary of any tender or exchange offer and to distribute or cause to be distributed to each Participant and Beneficiary such information as is distributed in connection with any tender or exchange offer to holders generally of Employer Securities, together with the appropriate forms for directing the Trustee as to the manner in which to respond to such tender or exchange offer. Upon timely receipt of directions under this Section 5.6 from the Participant or Beneficiary, the Trustee shall respond to the tender or exchange offer in accordance with, and only in accordance with, such directions. If the Trustee does not receive timely directions from a Participant or Beneficiary under this Section 5.6, the Trustee shall not tender, sell, convey or transfer any Employer Securities allocated to such person's Account in response to any tender or exchange offer.

5.7. <u>Investment Managers</u>. The Retirement Plan Committee may appoint one or more investment managers to manage all or any portion of all or any of the Investment Funds, and one or more custodians for all or any portion of any Investment Fund. The Retirement Plan Committee may also establish investment guidelines for the Trustee or any one or more investment managers and may direct that all or any portion of the assets in an Investment Fund be invested in one or more guaranteed investment contracts having such terms and conditions as the Retirement Plan Committee deems appropriate. The Retirement Plan Committee or the Trustee, at the direction of the Retirement Plan Committee, may enter into such agreements as the Retirement Plan Committee deems advisable to carry out the purposes of this Section 5.7.

5.8. <u>Valuations</u>. The Trust Fund shall be valued by the Trustee at fair market value as of the close of business on each Valuation Date. A similar valuation of the Trust Fund may be made at the end of any calendar month upon direction of the Retirement Plan Committee. In determining the fair market value of assets, the Trustee may appraise such assets itself or in its discretion, employ one or more appraisers for that purpose and rely on the values established by such appraiser or appraisers.

The Retirement Plan Committee or its delegate may, for administrative purposes, establish unit values for one or more Investment Funds (or any portion thereof) and maintain the accounts setting forth each Participant's interest in such Investment Fund (or any portion thereof) in terms of such units, all in accordance with such rules and procedures the Retirement Plan Committee or its delegate shall deem to be fair, equitable and administratively practicable. In the event that unit accounting is thus established for any Investment Fund (or any portion thereof) the value of a Participant's interest in that Investment Fund (or any portion thereof) at any time shall be an amount equal to the then-current value of a unit in such Investment Fund (or any portion thereof) multiplied by the number of units then credited to the Participant.

5.9. <u>Managed Account Service</u>. For purposes of this Article V, a Participant may act individually or utilize a managed account service, under which investment directions for the selection of investment options from the available Investment Funds (other than any Self-Managed Brokerage Account) will be provided by a registered investment manager on behalf of the Participant appointed under the Plan for this purpose, in accordance with rules and procedures established by the Retirement Plan Committee. Elections shall be subject to such additional rules or restrictions imposed by the Trustee or investment manager, and the Trustee and investment manager may decline to implement any election it deems inappropriate in light of such rules or restrictions.

ARTICLE VI ALLOCATIONS TO PARTICIPANT ACCOUNTS

6.1. <u>Participant Accounts</u>. The Trustee shall establish and maintain a Retirement Account, Pre-Tax Savings Account, Catch-Up Account, Post-Tax Savings Account, Matching Account, Rollover Account, Roth Savings Account and Roth Rollover Account, as appropriate, for each Participant. The Account shall be maintained solely for bookkeeping purposes and the Trustee shall not segregate the Plan's assets among the Accounts unless directed to do so.

6.2. Credits to and Charges Against Accounts. All crediting to and charging against Accounts shall be made as follows:

(a) First, the unit values for each Investment Fund shall be updated, dividends shall be posted and Participant Accounts shall be updated with the new unit values based upon the closing prices for each Investment Fund.

(b) Second, transfers shall be processed using the updated unit values.

(c) Third, all distributions and withdrawals made from the Accounts during the period from the prior Valuation Date to the current Valuation Date, and all expenses, if any, allocated to the Accounts, shall be charged against the Accounts using the updated unit values.

(d) Last, contributions shall be credited to each Participant's Account, using the updated unit values.

6.3. Expenses. All brokerage fees, transfer taxes, and other expenses incurred in connection with the investment of the Trust Fund shall be added to the cost of such investments or deducted from the proceeds thereof, as the case may be. All other costs and expenses of administering the Plan shall be paid from the Trust Fund unless the Employer elects to pay such costs and expenses. For the sake of clarity, the Retirement Plan Committee may allocate costs or expenses that are paid from the Trust Fund to Accounts on the basis it determines (e.g., per capita or pro rata). Notwithstanding the foregoing, the Retirement Plan Committee may direct that certain costs or expenses (e.g., loan fees, withdrawal fees, Investment Fund fees and lump sum distribution fees) shall be paid directly by or from the Accounts of those Participants who generate such costs or expenses, and the Retirement Plan Committee may direct that some or all of those costs or expenses shall be paid only by former Participants (directly or from their Accounts). Revenue sharing received by the Plan, if any, may be used to pay Plan administrative expenses or allocated as directed by the Retirement Plan Committee. Further, in the event that interest is generated on clearing accounts maintained by the Trustee, such funds shall be used to pay such Plan costs or expenses as the Retirement Plan Committee shall determine.

6.4. <u>Reimbursement of Trust Fund</u>. If the operation of the intra-plan accounting rules used in the administration of the Plan and/or the distribution provisions of the Plan result in a loss or charge against the Trust Fund or any portion of the Trust Fund, the Plan Sponsor may, in its sole discretion, make an additional contribution to the Plan to place the Trust Fund or the portion thereof in the position it would have been in with the absence of such loss or charge. Such a contribution shall be treated as a reimbursement of the Trust Fund and not as a contribution subject to Sections 4.1 through 4.7.

ARTICLE VII VESTING AND SEPARATION FROM SERVICE

7.1. Vested Percentage.

(a) A Participant shall at all times be fully vested in his or her Pre-Tax Savings Account, Post-Tax Savings Account, Rollover Account, Roth Rollover Account, Roth Savings



Account and Catch-Up Account. After December 31, 2001, a Participant shall also be fully vested in his or her Matching Account (including Matching Contributions made on or before December 31, 2001).

(b) A Participant's Retirement Account (and Matching Account prior to January 1, 2002), to the extent not previously vested under Subsection 7.1(c), shall become fully vested at the Participant's Normal Retirement Date, or upon his or her Total and Permanent Disability or death prior to otherwise incurring a Separation Date. The Normal Retirement Date, Total and Permanent Disability or death of a Participant after incurring a Separation Date shall not increase the vesting of the Participant's Account.

(c) A Participant's vested interest in the Participant's Retirement Account shall be determined under the following schedule.

Years of Vesting Service	Vested Percentage
less than 2	0%
2	20%
3	40%
4	60%
5	80%
6 or more	100%

(d) Notwithstanding anything in this Section 7.1 to the contrary, (i) if a Participant dies on or after January 1, 2007 while performing qualified military service (as defined in Code section 414(u)), the Participant's Retirement Account shall become fully vested and nonforfeitable upon the date of his death, provided that the Participant would have been eligible for reinstatement of employment with the Employer had the Participant's qualified military service ended on the day before his death, and (ii) if a Participant's employment is transferred from an Employer to TSAM (Delaware) LLC, prior to October 31, 2011, in connection with the initial formation of TSAM (Delaware) LLC, the Participants' Retirement Account shall become fully vested and nonforfeitable on the date of such transfer.

(e) Notwithstanding any provision of the Plan to the contrary, an ADS Transferred Associate shall become fully vested in his or her Retirement Account as of his or her ADS Transfer Date.

(f) Notwithstanding any provision of the Plan to the contrary, a Participant who is a Transferred Employee (as defined in Section 9.01 of that certain Asset Purchase Agreement dated as of December 12, 2018, by and among La Senza International, LLC, L Brands, Inc. and Regent, LP (the "Regent Purchase Agreement")) shall become fully vested in his or her Retirement Account as of his or her Transfer Date (as defined in Section 9.01 of the Regent Purchase Agreement).

(g) Notwithstanding any provision of the Plan to the contrary, a Participant whose employment is terminated by an Employer during the period commencing September 13, 2018 and ending February 2, 2019 as a direct result of the closure of L Brands' Henri Bendel business shall become fully vested in his or her Retirement Account as of January 1, 2019.

(h) Notwithstanding any provision of the Plan to the contrary, a Participant whose employment is involuntarily terminated in connection with the closure of a Victoria's Secret store occurring during the period commencing June 1, 2020 and ending December 31, 2020, shall become fully vested in his or her Retirement Account as of the date of his or her employment termination.

(i) Notwithstanding any provision of the Plan to the contrary, a Participant whose employment is involuntarily terminated in connection with a reduction in force announced by the Employer on July 28, 2020, shall become fully vested in his or her Retirement Account as of the date of his or her employment termination.

7.2. Forfeiture and Restoration. A Participant's Account may be forfeited, and any forfeited amounts restored, in accordance with this Section 7.2.

(a) A Participant shall experience a forfeiture of the unvested portion of his or her Account as follows:

(i) If a Participant has no vested interest in his or her Retirement Account and he or she incurs a Separation Date, he or she shall be deemed to have received a distribution of his or her Account and such Participant shall have the unvested portion of his or her Account forfeited as soon as administratively practicable following such Separation Date.

(ii) If a Participant is partially vested in his or her Retirement Account (i.e., has a vested percentage of more than 0%, but less than 100%), and he or she incurs a Separation Date, he or she shall have the unvested portion of his or her Account forfeited as soon as administratively practicable following the earlier of (A) the date on which he or she receives a distribution of the vested portion of his or her Account and (B) the first anniversary of the Participant's Separation Date.

(iii) If the Retirement Plan Committee determines that a Participant or Beneficiary is missing as described in Section 8.7 of the Plan, such Participant's or Beneficiary's benefit payment shall be forfeited as soon as administratively practicable following the determination that he or she is missing.

(b) Any amount that becomes a forfeiture as described in Subsection 7.2(a) shall be made available for the following purposes: (i) to reinstate previously forfeited account balances in accordance with Subsection 7.2(c) below; (ii) to reduce future Employer contributions, including, but not limited to, Retirement Contributions and Matching Contributions; (iii) to pay the costs and expenses of administering the Plan; and (iv) to reduce any additional amounts the Employer is responsible for under the Plan.

(c) Subject to the provisions of this Section 7.2(c), if a Participant or former Participant whose Account has been forfeited in whole or in part (the "Forfeited Participant") again becomes an active Participant prior to incurring five consecutive Breaks in Service, all amounts forfeited shall be restored (unadjusted for any gains or losses occurring after the last Valuation Date coinciding with or preceding the date of such forfeiture) to such Forfeited Participant's Account in accordance with this Subsection 7.2(c). If a Forfeited Participant

received a distribution of the vested portion of his or her Account, such Forfeited Participant shall not be entitled to restoration under this Subsection 7.2(c) unless and until he or she repays to the Plan the vested portion of his or her Account which was distributed (unadjusted for any gains or losses) before the earlier of: (i) five years after the first date on which such Forfeited Participant is subsequently reemployed by the Employer; or (ii) the close of the first period of five consecutive Breaks in Service commencing after the distribution. A restoration made under this Section 7.2(c) shall be made before the end of the Plan Year in which the Forfeited Participant once again becomes an active Participant or as soon thereafter as administratively practicable, or if the Forfeited Participant is required to repay the vested Account balance which was distributed, before the end of the Plan Year following the Plan Year in which such Forfeited Participant makes the required repayment. Restorations under this Section 7.2(c) shall be made from the forfeitures for the Plan Year of restoration (or, if no such forfeitures occur during that Plan Year, from additional Employer contributions).

(d) The vested portion of the Retirement Account of a Forfeited Participant who again becomes an active Participant shall be determined in a manner consistent with Regulation Section 1.411(a)-7(d)(5)(iii)(A) under the Code, with appropriate modifications to reflect forfeitures or partial forfeitures from such Account and the restoration of principal amounts of forfeitures.

7.3. Effect of Breaks in Service. If a person incurs five consecutive Breaks in Service and subsequently becomes a Participant, and if the Participant was not vested in any portion of his Account prior to the Breaks in Service, then the Years of Vesting Service completed by the Participant prior to such Breaks in Service shall not be taken into account in determining the Participant's vested interest in his or her Account accruing after the Breaks in Service. If a Participant was partially vested in his Account prior to incurring five consecutive Breaks in Service, upon incurring five or more consecutive Breaks in Service, the Participant shall forfeit any unvested portion of his or her Account that was accrued prior to such Breaks in Service and the Participant's vested interest in benefits accrued after such Breaks in Service shall be determined by taking into account any Years of Vesting Service accrued by the Participant before the Breaks in Service and any additional Years of Vesting Service accrued following the Breaks in Service. If any Years of Vesting Service are not required to be taken into account because of the operation of this Section 7.3, such Years of Vesting Service shall not be taken into account Breaks in Service.

7.4. <u>Amendments to Vesting Schedule</u>. If the vesting schedule under the Plan is amended, (a) no such amendment shall decrease any Participant's vested percentage, and (b) each Participant who has completed at least three Years of Vesting Service with the Employer may irrevocably elect in writing to have the vested percentage of his or her Account determined without regard to such amendment. The election period shall begin on the date the amendment is adopted and end on the latest of (i) the date 60 days after the Plan amendment is adopted; (ii) the date 60 days after the Participant is issued written notice of the Plan amendment by the Retirement Plan Committee or the Employer; or (iii) such later date as may be specified by the Retirement Plan Committee.

ARTICLE VIII BENEFITS

8.1. Benefit Payments after Separation Date.

(a) Except as otherwise elected in accordance with Section 8.2(h), a Participant or Beneficiary shall receive any benefits to which he or she is entitled in the form of a single-sum distribution consisting of cash, Employer Securities, and/or in-kind distributions from a Self-Managed Brokerage Account. Amounts not invested in either Employer Stock or in a Self-Managed Brokerage Account shall be distributed in cash. Amounts invested in an Employer Securities Contribution Fund shall be distributed in cash, unless the Participant or Beneficiary elects to receive the Employer Securities Contribution Fund shall be distributed in cash, unless the Participant or Beneficiary elects to receive the Employer Securities Contribution Fund in whole shares of the respective Employer Securities (plus cash for any fractional shares). Amounts invested in a Self-Managed Brokerage Account shall be distributed in cash, unless the Participant or Beneficiary elects to receive an in-kind distribution of those investments which can be taken in-kind and the remaining investments in cash; provided, however, that any partial distribution elected in accordance with Section 8.2(h) of amounts invested in a Self-Managed Brokerage Account shall be distributed solely in cash.

(b) Notwithstanding any provision of the Plan to the contrary that would otherwise limit a Distributee's election under this Section 8.1, a Distributee may elect, at the time and in the manner prescribed by the Retirement Plan Committee, to have any portion of an Eligible Rollover Distribution paid directly to an Eligible Retirement Plan specified by the Distributee in a Direct Rollover. For purposes of this Subsection 8.1(b), the following definitions shall apply:

(i) Eligible Rollover Distribution. An Eligible Rollover Distribution is any distribution of all or any portion of the balance to the credit of the Distributee, except that an Eligible Rollover Distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the Distributee or the joint lives (or joint life expectancies) of the Distributee and the Distributee's designated beneficiary, or for a specified period of ten years or more; any distribution to the extent such distribution is required under Section 401(a)(9) of the Code; the portion of any distribution that is not includable in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities); and any hardship distribution described in Section 401(k)(2)(B)(i)(IV) of the Code. A portion of a distribution shall not fail to be an Eligible Rollover Distribution merely because such portion consists of after-tax employee contributions which are not includible in gross income, provided that such portion may be transferred only to an individual retirement account or annuity described in § 408(a) or (b) of the Code, or to a qualified defined contribution plan described in § 401(a) or 403(a) of the Code that agrees to separately account for amounts so transferred, including separately accounting for the portion of such distribution which is includible in gross income and the portion of such distribution which is not so includible. Effective for distributions made on or after January 1, 2008, an Eligible Retirement Plan shall include a Roth IRA (as defined in Section 408A of the Code), subject to the restrictions set forth in Section 408A of the Code, which are incorporated herein by reference. A portion of a

distribution shall not fail to be an Eligible Rollover Distribution merely because such portion consists of Roth contributions, provided that such portion may be transferred only to a Roth IRA described in Section 408A of the Code, or to a designated Roth account within a qualified defined contribution plan as described in Section 402A of the Code.

(ii) <u>Eligible Retirement Plan</u>. An Eligible Retirement Plan is a plan which accepts the Distributee's Eligible Rollover Distribution and which is (A) an individual retirement account described in Section 408(a) of the Code, (B) an individual retirement annuity described in Section 408(b) of the Code, (C) a Roth IRA described in Section 408A of the Code, (D) an annuity plan described in Section 403(a) of the Code, (E) a qualified trust described in Section 401(a) of the Code, (F) an annuity contract described in Section 403(b) of the Code, or (G) an eligible plan under Section 457(b) of the Code that is maintained by a state, political subdivision of a state, or any agency or instrumentality of a state or political subdivision of a state and that agrees to separately account for amounts transferred into such plan from the Plan. The definition of Eligible Retirement Plan also shall apply with respect to any distribution made to a surviving Spouse, or to a Spouse or former Spouse who is the alternate payee under a qualified domestic relation(s) order, as defined in Section 414(p) of the Code.

(iii) <u>Distributee</u>. A Distributee includes an Associate or former Associate. In addition, the Associate's or former Associate's surviving Spouse and the Associate's or former Associate's Spouse or former Spouse who is the alternate payee under a qualified domestic relations order, as defined in Section 414(p) of the Code, are Distributees with regard to the interest of the Spouse or former Spouse. A non-Spouse Beneficiary shall be a Distributee under this Section 8.1(b)(iii) provided that the recipient Eligible Retirement Plan is an individual retirement account or individual retirement annuity and the distribution otherwise meets the requirements of Code Section 402(c)(11).

(iv) Direct Rollover. A Direct Rollover is a payment by the Plan to the Eligible Retirement Plan specified by the Distributee.

(c) On a periodic basis, the Retirement Plan Committee shall direct the person or entity maintaining Plan records to:

(i) determine all Participants who have incurred a Separation Date, who have not requested an Eligible Rollover Distribution as described in Section 8.1(b), and whose vested interest in his or her Account is \$1,000 or less as of the Valuation Date used for such determination; and to

(ii) distribute the entire vested balance in all such Participants' accounts in a single-sum distribution in accordance with Section 8.1(a) to such Participants as soon as administratively practicable after such Valuation Date.

(d) On any business day, the Retirement Plan Committee shall direct the person or entity maintaining Plan records to determine an interim value, and the vesting as of the preceding Valuation Date, of a Participant's Account for which a complete, single-sum distribution request has been made or which otherwise may be payable. As soon as administratively practicable after

the receipt of direction from the Retirement Plan Committee, the person or entity maintaining Plan records shall provide the information to the Trustee. As soon as administratively practicable after the receipt of information, the Trustee shall distribute any benefit that is payable to a Participant or Beneficiary. The number of shares of Employer Securities to be distributed to a Participant or Beneficiary shall be the number of shares in the Participant's Account as of the preceding Valuation Date, multiplied by the Participant's vested percentage as of the preceding Valuation Date. The amount of cash to be distributed to a Participant's Account not being distributed in Employer Securities, multiplied by the Participant's vested percentage as of the Valuation Date following a complete, single-sum distribution, the Retirement Plan Committee determines that a Participant's vested percentage increased and/or that additional contributions are allocable to the Participant's Account, a distribution shall be made of the additional distributable amount in accordance with this Section 8.1.

8.2. Retirement Benefit.

(a) Any Participant who has incurred a Separation Date shall receive a retirement benefit in an amount equal to the vested portion of the undistributed balance of the Participant's Account. If the value of the Participant's vested interest in his or her Account exceeds \$1,000, then the Participant shall receive his or her benefit after the earlier of the date the Participant requests a distribution or the date distribution is otherwise required to be made pursuant to Section 8.2(b) or (c).

(b) The distribution of the vested portion of the Participant's Account shall be made not later than 60 days after the close of the Plan Year in which the latest of the following occurs: (i) the Participant reaches his or her Normal Retirement Date; (ii) the Participant's 10th anniversary of participation in the Plan; or (iii) the Participant's Separation Date, unless a Participant elects to receive his or her benefit on a later date (but no later than the date specified in Subsection 8.2(c)).

(c) Notwithstanding any Plan provision to the contrary, all distributions required under the Plan shall be determined and made in accordance with Section 401(a)(9) of the Code, including the incidental death benefit requirement of Section 401(a)(9)(G) of the Code; Treasury Regulations Sections 1.401(a)(9)-2 through 1.401(a)(9)-9; and any revenue rulings, notices, and other guidance with respect to Section 401(a)(9) of the Code published in the Internal Revenue Bulletin. Accordingly, benefits must commence no later than the April 1 of the calendar year following the later of (i) the year in which the Participant attains age 70½ or (ii) the year in which the Participant retires. Paragraph (ii) of this Subsection 8.2(b) shall not apply, except as provided in Section 409(d) of the Code, in the case of a Participant who is a 5% owner (as defined in Section 416 of the Code) with respect to the Plan Year ending in the calendar year in which the Participant attains age 70½, or for purposes of Sections 408(a)(6) or (b)(3) of the Code. The amount to be distributed each calendar year may be any amount requested by the Participant which does not exceed the balance in his or her account, but must be at least an amount equal to the quotient obtained by dividing the Participant and his or her designated Beneficiary. Life expectancy and joint and last survivor expectancy are determined in accordance with Treasury Regulations

Section 1.401(a)(9)-9. This Section 8.2(c) specifically overrides any distribution option in the Plan inconsistent with Section 401(a)(9) of the Code.

(d) A Participant's election shall be made on a distribution request form provided by the Retirement Plan Committee and shall be effective if delivered to the Retirement Plan Committee by such date, before or after the applicable Valuation Date, as the Retirement Plan Committee establishes for purposes of administrative convenience.

(e) No distribution shall be made to any Participant before his or her Normal Retirement Date unless (i) the prior written consent of the Participant to the distribution has been obtained within the 90-day period ending on the date payments are to be made or commenced, or (ii) the value of the vested portion of the Participant's Accounts is \$1,000 or less as of the applicable Valuation Date.

(f) The Retirement Plan Committee shall notify the Participant of the right to defer any distribution until the Participant's Normal Retirement Date and the consequences of the failure to defer payment. Such notice shall include such information as needed to satisfy the notice requirements of Section 417(a)(3) of the Code and shall be provided no less than 30 days and no more than 180 days prior to the date that benefit payments are to be made.

If a distribution is one to which Sections 401(a)(11) and 417 of the Code do not apply, such distribution may commence less than 30 days after the notice required pursuant to this Subsection 8.2(f) is given, provided that (i) the Retirement Plan Committee clearly informs the Participant that the Participant has a right to a period of at least 30 days after receiving the notice to consider the decision of whether to elect a distribution; and (ii) the Participant, after receiving the notice, affirmatively elects a distribution.

(g) <u>Suspension for 2009</u>. Notwithstanding anything in Section 8.2(c) of the Plan to the contrary, a Participant or designated Beneficiary who would have been required to receive required minimum distributions for 2009 but for the enactment of Section 401(a)(9)(H) of the Code, and who would have satisfied that requirement by receiving "Extended 2009 RMDs" will not receive those distributions for 2009 unless the Participant or designated Beneficiary chooses to receive such distributions ("Extended RMD Participants"). For purposes of this Section 8.2(g), Extended 2009 RMDs means distributions that (A) are required minimum distributions for 2009 but for the enactment of Section 401(a)(9)(H) of the Code and (B) are (1) equal to the 2009 required minimum distributions or (2) one or more payments in a series of substantially equal distributions (that include the 2009 required minimum distributions) made at least annually and expected to last for the life (or life expectancy) of the Participant, the joint lives (or joint life expectancy) of the Participant and the Participant's designated Beneficiary, or for a period of at least 10 years. Extended RMD Participants will be given the opportunity to elect to receive the Extended 2009 RMDs in accordance with the procedures established by the Plan Administrator In addition, notwithstanding Section 8.1(b) of the Plan, and solely for purposes of applying the direct rollover provisions of the Plan, if an Extended RMD Participant elects to receive an Extended 2009 RMD, the Extended 2009 RMD amount distributed will be treated as an eligible rollover distribution and such amount will be eligible for direct rollover.

(h) <u>Partial Distributions after Separation Date</u>. A Participant who has incurred a Separation Date and whose vested interest in his or her Account is greater than \$1,000 may elect, pursuant to procedures implemented by the Retirement Plan Committee, to receive one or more partial distributions, including a series of partial distributions to be paid over a specified installment period, of the value of the Participant's vested Account balance. A Participant receiving a series of partial distributions may elect to cancel future scheduled distributions in accordance with procedures implemented by the Retirement Plan Committee, partial distributions will be charged (i) against the Participant's Roth Savings Account and Roth Rollover Account on a pro-rata basis or (ii) against the subaccounts in the Participant's Account (other than the Roth Savings Account and Roth Rollover Account) on a pro-rata basis, according to the Participant's election.

8.3. <u>Death Benefit</u>. If a Participant dies before receiving a distribution of his or her retirement benefit, the Participant's Beneficiary shall receive a death benefit, in lieu of the retirement benefit, equal to the value of the Participant's undistributed Account. Such distribution shall be paid, in a single-sum distribution in the manner set forth in Section 8.1(a), as soon as practicable after the Retirement Plan Committee is notified of the Participant's death.

8.4. Beneficiary Designation.

(a) A married Participant may, with the consent of his or her Spouse, designate and from time to time change the designation of one or more Beneficiaries or contingent Beneficiaries to receive any death benefit. The designation and consent shall be made in such manner as is determined by the Retirement Plan Committee. Any designation of a Beneficiary other than the Participant's Spouse shall require the written consent of the Spouse, which written consent must be witnessed by a Plan representative or a notary public. An unmarried Participant, or a married Participant whose Spouse has abandoned him or her or cannot be located, may designate a Beneficiary or Beneficiaries without the consent of any other person, provided that such Participant first establishes to the satisfaction of the Retirement Plan Committee either that he or she has no Spouse or that his or her Spouse cannot be located. All records of Beneficiary designations shall be maintained by the Retirement Plan Committee.

(b) In the event that the Participant fails to designate a Beneficiary to receive a benefit that becomes payable under the provisions of this Section 8.4, or in the event that the Participant is predeceased by all designated primary and contingent Beneficiaries, then (i) if the Participant is survived by a Spouse, the death benefit shall be payable to the Participant's surviving Spouse who shall be deemed to be the Participant's designated beneficiary for all purposes under the Plan; or (ii) if the Participant is not survived by a Spouse, the death benefit shall be payable to such members of the following classes of takers as can be located following reasonable efforts by the Retirement Plan Committee, each class to take to the exclusion of all subsequent classes, and all members of each class to share equally unless otherwise indicated: (A) lineal descendants (including adopted children and stepchildren), per stirpes; (B) surviving parents; or (C) the Participant's estate.

8.5. In-Service Withdrawals.

(a) An active Participant may withdraw all or any part of the vested portion of his or her Account after the attainment of age 59½, subject to the \$500 minimum withdrawal restriction set forth in Subsection 8.5(d) below. Age 59½ withdrawals shall be paid in cash and/or whole shares of Employer Securities, as elected by the Participant, and shall be charged against the subaccounts within a Participant's Account in the following order, subject to any other ordering rules or procedures as are implemented by the Retirement Plan Committee:

(i) In the case of a Participant electing distribution from Roth subaccounts: (A) Roth Rollover Account; and (B) the Roth Savings Account, together with the portion of the Catch-Up Account attributable to Roth Savings Contributions; and

(ii) In the case of a Participant electing distribution from non-Roth subaccounts: (A) Rollover Account (excluding the portion attributable to after-tax Rollover Contributions); (B) Post-Tax Savings Account; (C) the portion of the Rollover Account attributable to after-tax Rollover Contributions; (D) Pre-Tax Savings Account, together with the portion of the Catch-Up Account attributable to Pre-Tax Savings Contributions; (E) Matching Account; (F) Retirement Account; and (G) any other qualified nonelective or qualified matching contributions allocated to a Participant's Account not otherwise charged against one of the foregoing subaccounts.

(b) An active Participant who is fully vested in his or her Retirement Account and who has not attained age 59½ may obtain an in service withdrawal from his or her Retirement Account (subject to the \$500 minimum withdrawal restriction set forth in Subsection 8.5(d) below) as follows:

(i) In service withdrawals may be taken only from that portion of a Participant's Retirement Account attributable to Retirement Contributions made before January 1, 2009 (the "Pre-2009 Retirement Contributions") and earnings on such Pre-2009 Retirement Contributions.

(ii) An in-service withdrawal may not be taken from that portion of a Participant's Retirement Account attributable to Retirement Contributions made on or after January 1, 2009 (the "Post-2008 Retirement Contributions") or any earnings on such Post-2008 Retirement Contributions.

(c) A Participant may at any time make an in service withdrawal of all or a portion of his or her Post-Tax Savings Account, Roth Rollover Account or Rollover Account (including the portion of his or her Rollover Account attributable to after-tax Rollover Contributions), subject to the \$500 minimum withdrawal restriction set forth in Subsection 8.5(d) below and further subject to any ordering rules or procedures as are implemented by the Retirement Plan Committee.

(d) A request for an in-service withdrawal shall be made in accordance with such procedures and on such forms as the Retirement Plan Committee may determine. An in-service withdrawal shall be paid in cash and/or whole shares of Employer Securities, as elected by the Participant, as soon as practicable after the Participant requests an in-service withdrawal. The

amount of any in-service withdrawal made to a Participant, whether paid in cash, Employer Securities, or a combination thereof, shall not be less than \$500.

(e) A Participant may make an in-service withdrawal which qualifies as a "hardship withdrawal" as follows:

(i) A Participant who has an immediate and heavy financial need and who has obtained all distributions, other than hardship distributions, currently available under the Plan may receive a hardship distribution from his or her vested Account balance. For purposes of the Plan, an immediate and heavy financial need is the need for money for:

A. expenses for or necessary to obtain medical care for the Participant or the Participant's Spouse, children, dependents (as defined in Section 152 of the Code, without regard to Sections 152(b)(1), (b)(2) and (d)(1)(B) of the Code), or the Participant's Beneficiary designated in accordance with Section 8.4(a) hereof that would be deductible under Section 213(d) of the Code (determined without regard to the limitations in Section 213(a) of the Code);

B. costs directly related to the purchase (excluding mortgage payments) of a principal residence of the Participant;

C. the payment of tuition, related educational fees, and room and board expenses, for up to the next 12 months of postsecondary education for the Participant or the Participant's Spouse, children, dependents (as defined in Section 152 of the Code, without regard to Sections 152(b)(1), (b)(2) and (d)(1)(B) of the Code), or the Participant's Beneficiary designated in accordance with Section 8.4(a) hereof;

D. the prevention of the eviction of the Participant from his or her principal residence or foreclosure on the mortgage on that residence;

E. burial or funeral expenses for the Participant's deceased parent, Spouse, children, dependents (as defined in Section 152 of the Code, without regard to Section 152(d)(1)(B) of the Code), or the Participant's Beneficiary designated in accordance with Section 8.4(a) hereof;

F. the repair of damage to the Participant's principal residence that would qualify for the casualty deduction under Section 165 of the Code (determined without regard to (i) whether the loss exceeds 10% of adjusted gross income or (ii) the provisions of Section 165(h)(5) of the Code);

G. expenses and losses (including loss of income) incurred by the Participant on account of a disaster declared by FEMA under the Robert T. Stafford Disaster Relief and Emergency Assistance Act, provided that the Participant's principal place of residence or principal place of employment at the time of the disaster was located in an area designated by FEMA for individual assistance with respect to the disaster; or

H. any other similar purpose determined by the Retirement Plan Committee to be necessary to the maintenance of the Participant, or the Participant's Spouse, children, or dependents.

(ii) The amount of the hardship distribution shall be the lesser of (A) the Participant's vested Account balance or (B) the amount of the immediate and heavy financial need (including amounts necessary to pay reasonably anticipated taxes and penalties on the hardship distribution). In addition, effective January 1, 2020, no hardship withdrawal pursuant to this Section 8.5(e) shall be permitted unless (x) the Participant provides the Retirement Plan Committee a representation in writing, or in such other form permitted by the Retirement Plan Committee, that the Participant has insufficient cash or other liquid assets reasonably available to satisfy the need, and (y) the Retirement Plan Committee does not have actual knowledge that is contrary to the representation. Notwithstanding the foregoing, a Participant may not make a withdrawal for a financial hardship in an amount less than \$500 unless his or her vested Account balance is less than or equal to the amount of the financial hardship, in which case the Participant may withdraw 100% of his or her vested Account balance. Hardship distributions shall be paid in cash and shall be charged against the subaccounts within the Participant's Account in the following order, subject to any other ordering rules or procedures as are implemented by the Retirement Plan Committee: (i) Roth Rollover Account (excluding the portion attributable to after-tax Rollover Contributions); (iv) Post-Tax Savings Account; (v) the portion of the Rollover Account attributable to after-tax Rollover Contributions; (vi) Pre-Tax Savings Account, together with the portion of the Catch-Up Account attributable to Pre-Tax Savings Contributions; (vii) Retirement Account; and (ix) any other qualified nonelective or qualified matching contributions allocated to a Participant's Account not otherwise charged against one of the foregoing subaccounts (if any).

(f) <u>Withdrawals During Qualified Military Service</u>. Notwithstanding the foregoing, to the extent permitted by and subject to the provisions of the Heroes Earnings Assistance and Relief Tax Act of 2008 and applicable law, a Participant who is performing service in the uniformed services (as defined in chapter 43 of title 38, United States Code) while on active duty for a period of at least 30 days and receiving differential wage payments (within the meaning of Code section 3401(h)(2)) shall be eligible to receive a distribution of his elective deferrals from his Account, provided that such Participant may not make Pre-Tax Savings Contributions, Roth Savings Contributions and Post-Tax Savings Contributions under the Plan during the six month period following such distribution. Any such distribution of a portion of the Participant's elective deferrals shall be charged to the Participant's Pre-Tax Savings Account according to the Participant's election, subject to any ordering rules or procedures as are implemented by the Retirement Plan Committee.

8.6. <u>Post Distribution Credits</u>. If, after the complete distribution of retirement or death benefits under the Plan, there remain in a Participant's Account any vested funds, or any vested funds shall be subsequently credited thereto, such funds shall be paid to the Participant or his or her Beneficiary as promptly as practicable in the same manner as the initial distribution.

8.7. <u>Lost Participants</u>. If the Retirement Plan Committee cannot ascertain the whereabouts of any person to whom a payment is due under the Plan, such payment may be treated as follows:

(a) If, prior to a payment and after the Retirement Plan Committee has made reasonable efforts to locate a missing Participant or Beneficiary, the Retirement Plan Committee is unable to locate the Participant or Beneficiary, then the payment due to such Participant or Beneficiary shall be forfeited in accordance with Section 7.2 of the Plan.

(b) If a payment has been made by issuing a check and such check is outstanding for more than 180 days, and the Retirement Plan Committee is unable to locate the Participant or Beneficiary who is the payee on the check after making reasonable efforts to do so, then the check and all remaining payments that would otherwise be due to such person shall be cancelled forfeited in accordance with Section 7.2 of the Plan.

(c) If a Participant or Beneficiary's payment is treated as a forfeiture under Subsections 8.7(a) or 8.7(b) above, and such Participant or Beneficiary is located subsequent to such forfeiture, the Retirement Plan Committee shall reinstate the forfeited amount without adjustment for any investment gains or losses which may have occurred after the date of the forfeiture. A reinstatement under this Subsection 8.7(c) shall not be treated as an Annual Addition under Section 415 of the Code or Section 4.6 of the Plan.

(d) If a Participant or Beneficiary's payment is treated as a forfeiture under Subsections 8.7(a) or 8.7(b) above, and the Plan is joined as a party to escheat proceedings involving such forfeited payment, the Plan will comply with the final judgment as if were a claim filed by the former Participant or Beneficiary and the Plan will make payment in accordance with the judgment.

(e) A Participant's or Beneficiary's payment shall not be treated as a forfeiture under Subsection 8.7(a) or 8.7(b) above unless and until the Retirement Plan Committee makes such efforts as are considered a reasonable fulfillment of the Retirement Plan Committee's fiduciary obligations under ERISA Section 404(a), including use of such search methods as are recommended in guidance issued by the Department of Labor. To the extent a Participant's or Beneficiary's payment is forfeited in accordance with this Section 8.7, such forfeited amount may be used as described in Section 7.2(b).

ARTICLE IX. THE RETIREMENT PLAN COMMITTEE

9.1. <u>Appointment and Tenure</u>. The Retirement Plan Committee shall be a committee of not less than three and not more than five members who are appointed by the Compensation Committee. The Retirement Plan Committee shall be the "named fiduciary" of the Plan (within the meaning of Section 402(a)(3) of ERISA), the "administrator" of the Plan (within the meaning of Section 3(16)(A) of ERISA) and the "plan administrator" of the Plan (within the meaning of Section 414(g) of the Code). Any Retirement Plan Committee member may be dismissed or may resign at any time, with or without cause, upon 10 days' written notice. Vacancies arising by

reason of the death, resignation, or removal of a Retirement Plan Committee member shall be filled by such officers or management-level Associates as may be designated by the Compensation Committee. If the Compensation Committee fails to act, and in any event until the Compensation Committee so acts, the remaining members of the Retirement Plan Committee may appoint an interim Retirement Plan Committee member to fill any vacancy occurring on the Retirement Plan Committee.

9.2. <u>Meetings; Majority Rule</u>. The Retirement Plan Committee may act by majority vote of those present taken in a meeting if all members of the Retirement Plan Committee have received at least 10 days' written notice of such meeting or have waived notice and a quorum of a majority of all members is present. The Retirement Plan Committee may also act by majority consent in writing without a meeting.

9.3. Delegation. The Retirement Plan Committee may delegate to any of its members or any other person the authority to sign any documents on its behalf or to perform ministerial acts, but no such member or person shall perform any act involving the exercise of any discretion without first obtaining the concurrence of a majority of the members of the Retirement Plan Committee, even though he or she alone may sign any document required by third parties. The Retirement Plan Committee may elect one of its members to serve as chairman. The chairman shall preside at all meetings of the Retirement Plan Committee or shall delegate such responsibility to another Retirement Plan Committee member. The Retirement Plan Committee may elect one or more persons to serve as secretary or assistant secretary of the Retirement Plan Committee. The secretary or assistant secretary may, but need not, be a member of the Retirement Plan Committee. All third parties may rely on any communication signed by the secretary or assistant secretary, acting as such, as an official communication from the Retirement Plan Committee.

9.4. <u>Appointment of Trustee</u>. The Retirement Plan Committee shall have the sole discretionary authority to select, monitor and remove the Trustee, subject to the provisions of the Trust Agreement.

9.5. <u>Investment Authority</u>. The Retirement Plan Committee shall have such powers and authorities regarding the investment of Plan assets as are set forth in ARTICLE V hereof. The Retirement Plan Committee shall also establish, and communicate to the Trustee, an investment policy for the Plan.

9.6. <u>Reporting and Disclosure</u>. The Retirement Plan Committee shall keep all individual and group records relating to Participants, former Participants, and Beneficiaries, and all other records necessary for the proper operation of the Plan, except that the Retirement Plan Committee shall have the right to exercise such authority while conducting any review of a denied claim for benefits. Such records shall be made available to the Employer and to each Participant and Beneficiary for examination during business hours. A Participant or Beneficiary may examine only such records as pertain exclusively to the examining Participant or his or her Beneficiary and the Plan and Trust Agreement. The Retirement Plan Committee shall prepare and furnish to Participants annual statements and all information required by applicable law or the Plan. The Retirement Plan Committee shall prepare and publish, or file with the appropriate governmental officials, all reports and other information required by law to be filed or published.

9.7. <u>Construction of the Plan</u>. The Retirement Plan Committee shall have sole discretionary authority to interpret the Plan and determine conclusively all questions arising in the administration, interpretation, and application of the Plan, including, but not by way of limitation, the power to determine, with sole discretionary authority, the rights or eligibility of Associates, Participants, and Beneficiaries, and the amount of their respective benefits. Decisions of the Retirement Plan Committee regarding claims shall be final and binding on all persons. The Retirement Plan Committee shall endeavor to act, whether by general rules or by particular decisions, so as to treat all persons in similar circumstances uniformly. The Retirement Plan Committee's interpretations and determinations shall be final and binding on all persons absent fraud or arbitrary and capricious abuse of the wide discretion granted to the Retirement Plan Committee. The Retirement Plan Committee shall, provide the Trustee with instructions regarding payments of benefits. The Retirement Plan Committee shall provide directions to the Trustee with respect to valuations at dates other than Valuation Dates and all other matters when called for in the Plan or requested by the Trustee. The Retirement Plan Committee may waive any period of notice required under the Plan. The Retirement Plan Committee shall provide procedures for the determination of claims for benefits, including procedures regarding the review of denied claims.

9.8. <u>Engagement of Assistants and Advisors</u>. The Retirement Plan Committee shall have the right to hire such professional assistants and consultants as it, in its sole discretion, deems necessary or advisable, including, but not limited to investment managers and/or advisors, accountants, actuaries, attorneys, consultants, clerical and office personnel, and medical practitioners. The costs for such assistants and advisors shall be paid from the Trust Fund as an expense of the Trust Fund unless the Employer elects to pay such costs and expenses.

9.9. <u>Bonding</u>. The Retirement Plan Committee shall arrange for such bonding as is required by law, but no bonding in excess of the amount required by law shall be required under the Plan.

9.10. <u>Compensation</u>. The members of the Retirement Plan Committee shall serve without compensation for their services, but all expenses of the members in connection with administering the Plan shall be paid or reimbursed by the Trust Fund, except to the extent paid by the Employer.

9.11. <u>Indemnification of the Retirement Plan Committee</u>. Each member of the Retirement Plan Committee shall be indemnified by the Employer against costs, expenses and liabilities (other than amounts paid in settlements to which the Employer does not consent) reasonably incurred by him or her in connection with any action to which he or she may be a party by reason of his or her service as a member of the Retirement Plan Committee except in relation to matters as to which he or she be adjudged in such action to be personally guilty of gross negligence or willful misconduct in the performance of his or her duties. The foregoing right to indemnification shall be in addition to such other rights as the Retirement Plan Committee member may enjoy as a matter of law or by reason of insurance coverage of any kind, or otherwise. Service as a Retirement Plan Committee member shall be deemed in partial fulfillment of the member's function as an associate, officer and/or director of the Employer, if he or she serves in such capacity as well.

ARTICLE X_ ALLOCATION OF AUTHORITY AND RESPONSIBILITIES

10.1. <u>Authority and Responsibilities of the Plan Sponsor</u>. The Plan Sponsor shall have sole discretionary authority within the limits provided by the Plan to do, or cause to be done by one or more parties appointed by the Plan Sponsor, the following:

(a) to appoint the Retirement Plan Committee and to monitor its performance;

(b) to communicate such information to the Retirement Plan Committee and the Trustee as each needs for the proper performance of its duties;

(c) to provide mechanisms through which the Retirement Plan Committee and the Trustee can communicate with Participants and Beneficiaries;

(d) to perform such duties as are imposed by applicable law and to serve as the Retirement Plan Committee in the absence of an appointed Retirement Plan Committee; and

(e) to carry out such other duties as may be assigned to it hereunder.

10.2. <u>Authority and Responsibilities of the Retirement Plan Committee</u>. The Retirement Plan Committee shall be the Named Fiduciary with respect to the authority and responsibilities described in ARTICLE IX. The Retirement Plan Committee may delegate certain of its non-fiduciary authority and responsibilities to the Plan's administrative services provider.

10.3. Authority and Responsibilities of the Trustee. The Trustee shall have the powers and duties set forth in the Trust Agreement.

10.4. <u>Limitations on Obligations</u>. The Retirement Plan Committee shall not have authority or responsibility to deal with matters other than as delegated to it under the Plan, under the Trust Agreement, or by operation of law. A Named Fiduciary shall not in any event be liable for breach of fiduciary responsibility or obligation by another fiduciary if the responsibility or authority of the act or omission deemed to be a breach was not within the scope of the said Named Fiduciary's authority or responsibility.

ARTICLE XI CLAIMS PROCEDURES

11.1. <u>Application for Benefits</u>. Each Participant or Beneficiary believing himself or herself eligible for benefits under the Plan may apply for such benefits by completing and filing with the Retirement Plan Committee an application for benefits in writing. Before the date on which benefit payments commence, each such application must be supported by such information and data, as the Retirement Plan Committee deems relevant and appropriate.

11.2. Appeals of Denied Claims for Benefits.

(a) Claims for benefits under the Plan shall be made in writing to the Retirement Plan Committee or its duly authorized delegate. If the Retirement Plan Committee or such delegate wholly or partially denies a claim for benefits, the Retirement Plan Committee or if applicable, its delegate shall, within a reasonable period of time, but no later than 90 days after receipt of the claim, notify the claimant in writing or electronically of the adverse benefit determination. Notice of an adverse benefit determination shall be written in a manner calculated to be understood by the claimant and shall contain: (i) the specific reason or reasons for the adverse benefit determination; (ii) a specific reference to the pertinent Plan provisions upon which the adverse benefit determination of why such material or information is necessary; and (iv) an explanation of the Plan's review procedure and the time limits applicable to such procedure, including a statement of the claimant's right to bring a civil action under Section 502(a) of ERISA following an adverse benefit determination. If the Retirement Plan Committee or its delegate determines that an extension of time is necessary for processing the claim, the Retirement Plan Committee or its delegate shall notify the claimant in writing of such extension, the special circumstances requiring the extension, and the date by which the Retirement Plan Committee expects to render the benefit determination. In no event shall the extension exceed a period of 90 days from the end of the initial 90-day period.

(b) Within 60 days after the claimant receives the written or electronic notice of an adverse benefit determination or such later time as shall be deemed reasonable in the sole discretion of the Retirement Plan Committee taking into account the nature of the benefit subject to the claim and other attendant circumstances, the claimant may file a written request with the Retirement Plan Committee that it conduct a full and fair review of the adverse benefit determination, including the holding of a hearing, if deemed necessary by the Retirement Plan Committee. In connection with the claimant's appeal of the adverse benefit determination, the claimant may review pertinent documents and may submit issues and comments in writing. The Retirement Plan Committee shall render a decision on the appeal promptly, but not later than 60 days after the receipt of the claimant's request for review, unless special circumstances (such as the need to hold a hearing, if necessary) require an extension of time for processing, in which case the 60-day period may be extended to 120 days. The Retirement Plan Committee shall notify the claimant in writing of any such extension, the special circumstances requiring the extension, and the date by which the Retirement Plan Committee expects to render the determination, such notice shall (i) include specific reasons for the adverse determination; (ii) be written in a manner calculated to be understood by the claimant; (iii) contain specific references to the pertinent Plan provisions upon which the benefit determination is based; (iv) contain a statement that the claimant is entitled to receive upon request and free of charge, reasonable access to, and copies of, all documents, records, and other information relevant to the claimant's claim for benefits; and (v) contain a statement of the claimant's right to bring an action under Section 502(a) of ERISA.

11.3. <u>Review of Decision</u>. Any final decision of the Retirement Plan Committee may be reviewed by a court of competent jurisdiction only if an appeal has been made under Section 11.2.

ARTICLE XII_ AMENDMENT, TERMINATION, MERGERS AND CONSOLIDATIONS

12.1. Amendment.

(a) The provisions of the Plan may be amended at any time and from time to time by the Board of Directors acting through the Compensation Committee; provided, however, that, except as otherwise permitted by law:

- (i) No amendment shall increase the duties or liabilities of the Trustee without the consent of the Trustee.
- (ii) No amendment shall decrease the balance in any Account.

(iii) No amendment shall provide for the use of funds or assets held to provide benefits under the Plan other than for the benefit of Associates and Beneficiaries, except as may be permitted by applicable law.

(b) Without limiting in any way the authority of the Compensation Committee pursuant to Section 12.1(a), the most senior executive with responsibility over human resources of L Brands (or, if such responsibility is shared among two or more executives, then any such executive) (an "Authorized Executive") may approve amendments to the Plan that, in the opinion of the Authorized Executive are either (i) legally required in order to maintain the Plan's qualified status under Code Sections 401(a) and 401(k) or to comply with the applicable requirements of ERISA or (ii) do not substantially increase the cost of funding or administering the Plan.

12.2. <u>Termination</u>. While it is the Plan Sponsor's intention to continue the Plan indefinitely in operation, the Plan Sponsor nevertheless reserves the right to terminate the Plan in whole or in part by resolution of its Board of Directors. Termination or partial termination of the Plan shall result in full and immediate vesting in each affected Participant of the affected portion of his or her Account Plan termination shall be effective as of the date specified by the Plan Sponsor. The Plan Sponsor shall instruct the Trustee on termination of the Plan either to continue to manage and administer the Trust Fund for the benefit of Participants and Beneficiaries pursuant to the terms and provisions of the Trust Agreement, or to pay over to each Participant (and deferred vested former Participant) the value of his or her vested interest, and to thereupon dissolve the Trust Fund. Notwithstanding the foregoing, except as permitted by Section 401(k) of the Code, no portion of a Participant's Pre-Tax Savings Account or Roth Savings Account shall be distributed prior to the Participant's retirement, death, disability, Separation Date, attainment of age 59½, or proven financial hardship.

12.3. <u>Permanent Discontinuance of Contributions</u>. While it is the Employer's intention to make substantial and recurring contributions to the Trust Fund under the Plan, the Plan Sponsor nevertheless reserves the right at any time to permanently discontinue all Employer contributions. Such permanent discontinuance shall be established by resolution of the Board of Directors and shall have the same effect as a termination of the Plan, except that the Trustee shall not have the authority to dissolve the Trust Fund except upon adoption of a further resolution by

the Board of Directors to the effect that the Plan is terminated and upon receipt from the Plan Sponsor of instructions to dissolve the Trust Fund.

12.4. <u>Suspension of Employer Contributions</u>. The Plan Sponsor shall have the right, at any time and from time to time, to suspend contributions to the Trust Fund. Such suspension shall have no effect on the operation of the Plan except as set forth below:

(a) If the Plan Sponsor determines that such suspension shall be permanent, a permanent discontinuance of contributions will be deemed to have occurred as of the date of such determination or such earlier date as is specified.

(b) If a temporary suspension becomes a Plan termination or a complete discontinuance, the termination shall be deemed to have occurred on the earlier of the date specified by the Plan Sponsor or established by the Retirement Plan Committee or the last day of the Plan Year next following the first Plan Year during the period of suspension in which the Plan Sponsor did not make contributions.

12.5. <u>Mergers and Consolidations of Plans</u>. In the event of any merger or consolidation with, or transfer of assets or liabilities to, any other plan, each Participant, deferred vested former Participant and Beneficiary shall have a benefit in the surviving or transferee plan (determined as if such plan were then terminated immediately after such merger) that is equal to or greater than the benefit he or she would have been entitled to receive immediately before such merger in the Plan in which he or she was then a Participant or with respect to which he or she was then a deferred vested former Participant or Beneficiary (had such Plan been terminated at that time).

12.6. <u>Transfers of Assets to or from the Plan</u>. A transfer of all or any portion of the assets or liabilities of the Plan to any other plan, or the transfer of all or any portion of the assets or liabilities of another plan to the Plan, shall be on the terms and conditions set forth in (i) Section 414(l) of the Code; (ii) resolutions of the Compensation Committee, to the extent not inconsistent with (i); and (iii) a duly executed agreement approved or authorized by the Compensation Committee concerning such a transfer, to the extent not inconsistent with (i) or (ii). The Plan shall not accept a direct or indirect transfer of assets that would make the Plan subject to Sections 401(a)(11) and 417 of the Code with respect to any Participant.

ARTICLE XIII PARTICIPATING EMPLOYERS

13.1. <u>Adoption by Other Entities</u>. With the consent of the Plan Sponsor, any other entity, whether or not an Affiliate, may adopt the Plan and all of the provisions hereof as to all or any category of its associates, as a participating Employer, by a properly executed document evidencing the intent and will of the board of directors of the other corporation.

13.2. <u>Requirements of Participating Employers</u>. Each participating Employer shall be required to use the same Trustee and Trust Agreement as provided in the Plan, and the Trustee shall commingle, hold and invest as the Trust Fund all contributions made by participating Employers, as well as all increments thereof.

13.3. <u>Designation of Agent</u>. With respect to all relations with the Trustee and Retirement Plan Committee, each participating Employer shall be deemed to have irrevocably designated the Plan Sponsor as its agent.

13.4. <u>Discontinuance of Participation</u>. Any participating Employer may discontinue or revoke its participation in the Plan. At the time of any such discontinuance or revocation, satisfactory evidence thereof and of any applicable conditions imposed shall be delivered to the Trustee. The Trustee shall retain assets for the Associates of the discontinued Employer under the Plan.

13.5. <u>Retirement Plan Committee's Authority</u>. The Retirement Plan Committee shall have authority to make any and all necessary rules or regulations, binding upon all participating Employers and all Participants, to effectuate the purposes of the Plan.

ARTICLE XIV_ MISCELLANEOUS PROVISIONS

14.1. Nonalienation of Benefits.

(a) None of the payments, benefits, or rights of any Participant or Beneficiary shall be subject to any claim of any creditor, and, in particular, to the fullest extent permitted by law, all such payments, benefits, and rights shall be free from attachment, garnishment, trustee's process, or any other legal or equitable process available to any creditor of such Participant or Beneficiary, and no Participant or Beneficiary shall have the right to alienate, anticipate, commute, pledge, encumber, or assign any of the benefits or payments which he or she may expect to receive, contingently or otherwise, under the Plan, except the right to designate a Beneficiary or Beneficiaries as hereinbefore provided.

Notwithstanding the foregoing, for judgments, orders and decrees issued, and settlements, a Participant's benefit in the Plan may be reduced to satisfy liabilities of the Participant to the Plan due to:

(i) the Participant's conviction of a crime involving the Plan;

(ii) a civil judgment (or consent order or decree) entered by a court or an action brought in connection with a violation of the fiduciary provisions of ERISA; or

(iii) a settlement agreement between the Secretary of Labor and the Participant in connection with a violation of the fiduciary provisions of ERISA.

To be effective, the court order establishing such liability must require that the Participant's benefit in the Plan be applied to satisfy the liability. Spousal consent is required to offset the liability, unless the court order also requires the Spouse to pay an amount to the Plan.

(b) The foregoing shall not prohibit distributions under a "qualified domestic relations order" as defined in Section 414(p) of the Code. The Retirement Plan Committee shall establish a written procedure to determine the qualified status of domestic relations orders and to

administer distributions under such qualified orders and in accordance with Section 414(p) of the Code. A former Spouse of a Participant shall be treated as his or her Spouse under the Plan only to the extent specified under a qualified domestic relations order.

(c) In addition, the Plan specifically authorizes distributions to an alternate payee under a qualified domestic relations order regardless of whether the Participant has attained the earliest retirement age (as defined in the preceding sentence and in Section 414(p) of the Code) only if: (i) the order specifies distribution at the earlier date or permits an agreement between the Plan and the alternate payee authorizing an earlier distribution; and (ii) the alternate payee consents to a distribution prior to the Participant's earliest retirement age if the present value of the alternate payee's benefits under the Plan exceeds \$1,000.

14.2. <u>No Contract of Employment</u>. Neither the establishment of the Plan, nor any modification thereof, nor the creation of any fund, trust or Account, nor the payment of any benefits, shall give any Participant, Associate, or other person whosoever the right to be retained in the service of the Plan Sponsor, an Affiliate or any other participating Employer, and all Participants and other Associates shall remain subject to discharge to the same extent as if the Plan had never been adopted.

14.3. <u>Severability</u>. If any provision of the Plan shall be held invalid or unenforceable, such invalidity or unenforceability shall not affect any other provision hereof, and the Plan shall be construed and enforced as if such invalid or unenforceable provision had not been included.

14.4. <u>Successors</u>. The Plan shall be binding upon the heirs, executors, administrators, personal representatives, successors, and assigns of the parties, including each Participant and Beneficiary, present and future.

14.5. <u>Captions</u>. The headings and captions herein are provided for convenience only, shall not be considered a part of the Plan, and shall not be employed in the construction of the Plan.

14.6. <u>Gender and Number</u>. Except where otherwise clearly indicated by context, the masculine gender shall include the feminine gender, the singular shall include the plural, and vice versa.

14.7. <u>Controlling Law</u>. The Plan shall be construed and enforced according to the laws of the State of Delaware to the extent not preempted by federal law, which shall otherwise control.

14.8. <u>Title to Assets</u>. No Participant or Beneficiary shall have any right to, or interest in, any assets of the Trust Fund, upon termination of his or her employment or otherwise, except to the extent of the benefits payable under the Plan to such Participant out of the assets of the Trust Fund. All payments of benefits under the Plan shall be made from the assets of the Trust Fund, and neither the Employer nor any other person shall be liable therefor in any manner.

14.9. <u>Payments to Minors, Etc</u>. Any benefit payable to or for the benefit of a minor, an incompetent person or other person incapable of receipting therefor shall be deemed paid when paid to such person's guardian, to a trustee holding assets for such person or to the party

providing, or reasonably appearing to provide, for the care of such person, and such payment shall fully discharge the Trustee, the Retirement Plan Committee, the Employer and all other parties with respect thereto.

14.10. <u>Risk to Participants</u>. Each Participant assumes all risks associated with any decrease in the value of any securities in the Trust Fund and agrees that the Trust Fund shall be the sole source of payments under the Plan and none of the Plan Sponsor, any Affiliate, any other participating Employer, or any member of the Retirement Plan Committee shall be responsible for the payment of any benefits under the Plan.

14.11. Entire Agreement; Successors. The Plan, including any election agreements and any amendments thereto, shall constitute the entire agreement between the Plan Sponsor or any Employer and the Participant or any Beneficiary. No oral statement regarding the Plan may be relied upon by the Participant or any Beneficiary. The Plan and any amendment shall be binding on the parties thereto and their respective heirs, administrators, trustees, successors and assigns, and on all Participants and Beneficiaries. By becoming a Participant, each Associate shall be conclusively deemed to have assented to the provisions of the Plan and the Trust Agreement and to any amendments thereto.

14.12. <u>Approval by Internal Revenue Service</u>. Notwithstanding anything herein to the contrary, if the Commissioner of Internal Revenue Service or his or her delegate should determine that the Plan, as amended and restated, does not qualify as a tax-exempt plan and trust under Sections 401 and 501 of the Code, and such determination is not contested, or if contested is finally upheld, then the Plan shall operate as if it had not been amended and restated.

14.13. <u>Electronic and Telephonic Elections</u>. Notwithstanding any provision in the Plan to the contrary, salary reduction agreements and cancellations or amendments thereto, investment elections, changes or transfers, loans, withdrawals decisions, and any other decision or election by a Participant (or Beneficiary) under the Plan may be accomplished by electronic or telephonic means that are not otherwise prohibited by law and that are in accordance with procedures and/or systems approved or arranged by the Retirement Plan Committee or its delegates.

14.14. <u>Special Provisions for Participants Who Enter the Armed Forces</u>. Notwithstanding any other provision of the Plan to the contrary, effective for reemployment initiated on or after December 12, 1994, contributions, benefits and service credit will be provided in accordance with Section 414(u) of the Code.

14.15. <u>Correction of Mistakes</u>. If the Retirement Plan Committee determines that a mistake occurred in the administration of the Plan, the Retirement Plan Committee shall ensure that such mistake is corrected in accordance with the Plan terms as interpreted by the Retirement Plan Committee in its sole discretion, or as otherwise required by law, including under applicable Internal Revenue Service and Department of Labor correction procedures.

14.16. <u>Restriction on Venue and Limitations on Actions</u>. Any Participant or Beneficiary can bring an action in connection with the Plan only in Federal District Court in Columbus,

Ohio. The Participant or Beneficiary must bring the cause of action within two years of the initial adverse benefit determination.

* * *

IN WITNESS WHEREOF, the Plan Sponsor has caused the Plan to be executed by its duly authorized officer this 26th day of March, 2021.

L BRANDS SERVICE COMPANY, LLC

By: /s/ Wendy Arlin

Title: SVP, Corporate Controller and Finance

FIRST AMENDMENT TO THE L BRANDS, INC. 401(k) SAVINGS AND RETIREMENT PLAN (AS AMENDED AND RESTATED EFFECTIVE APRIL 5, 2021)

WHEREAS, L Brands Service Company, LLC, formerly known as Limited Brands Service Company, LLC (the "Company") sponsors the L Brands, Inc. 401(k) Savings and Retirement Plan, as amended and restated effective April 5, 2021 (the "Plan"), for the benefit of eligible associates of the Company and its affiliates; and

WHEREAS, the Company desires to amend the Plan in connection with the anticipated Spinoff (as defined below) of Victoria's Secret & Co. and its subsidiaries (i) to reflect that employees of Victoria's Secret & Co. and its subsidiaries will no longer be eligible to make contributions to the Plan or receive contributions under the Plan as of the Spinoff Date (as defined below), (ii) to reflect the related trust-to-trust transfer of assets and liabilities from the Plan to the Victoria's Secret & Co. 401(k) Savings and Retirement Plan, (iii) to provide credit for service with Victoria's Secret & Co. and its subsidiaries in the case of Delayed LB Transfer Employees (as defined below) returning to service with the Company and its affiliates after the Spinoff, (iv) to provide for full vesting in the case of Delayed VS Transfer Employees (as defined below) terminating employment with the Company and its affiliates following the Spinoff, (v) to prospectively eliminate in-kind rollover contributions, (vi) to set forth Plan terms applicable to shares of common stock of Victoria's Secret & Co. held in the Plan as a result of the Spinoff, (vii) to permit in-service withdrawals by employees of Victoria's Secret & Co. and its subsidiaries during the transition period which shall commence immediately following the Spinoff and end upon the 401(k) Transfer Date (as defined below), (viii) to reflect the name change of L Brands, Inc. to Bath & Body Works, Inc. and to reflect the change in the name of the Plan from the L Brands, Inc. 401(k) Savings and Retirement Plan; and (ix) to make certain other changes as described more fully below;

NOW, THEREFORE, BE IT RESOLVED, effective as of August 2, 2021, unless otherwise set forth below, the Plan is hereby amended as follows:

1. Section 1.1 of the Plan is hereby amended by inserting a new paragraph at the end thereof to read as follows:

"Prior to August 2, 2021 (the 'Spinoff Date'), VS Service Company, LLC ('VS') and Victoria's Secret & Co. ('VS Parent') were wholly owned subsidiaries of L Brands, Inc. On the Spinoff Date, L Brands, Inc. distributed to its stockholders all of the outstanding shares of common stock, par value \$0.01 per share, of VS Parent (the 'Spinoff'), and as a result VS Parent and VS ceased to be subsidiaries of L Brands, Inc. In connection with the Spinoff and subject to the terms of the Employee Matters Agreement, the Transferred Participants, as defined in Article II, will cease active participation in this Plan and

become eligible to participate in the Victoria's Secret & Co. 401(k) Savings and Retirement Plan sponsored by VS (the 'VS Plan') as of the Spinoff Date, and the assets and liabilities of the Plan attributable to the Transferred Participants will be spun off from this Plan to the VS Plan, effective as of the close of business on August 31, 2021 (the '401(k) Transfer Date'). In addition, effective as of the Spinoff, L Brands, Inc. shall change its name to Bath & Body Works, Inc. ('BBWI'). The Plan is hereby amended to reflect the change in the name of the Plan from the L Brands, Inc. 401(k) Savings and Retirement Plan to the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan effective as of the Spinoff Date."

2. Clause (v) of Section 2.6 of the Plan is hereby amended to read as follows:

"(v) employed by Puerto Rico Store Operations, LLC (formerly known as Victoria's Secret Stores Puerto Rico, LLC), which is an Affiliate of the Employer."

3. The reference to "Sections 2.26, 4.6 and 4.11" in Section 2.13 of the Plan is hereby deleted and replaced with a reference to "Sections 2.29, 4.6 and 4.11."

4. Article II of the Plan is hereby amended by (i) inserting new Sections 2.15 and 2.16 immediately following existing Section 2.14 to read as follows and (ii) renumbering the succeeding sections of Article II accordingly:

"2.15. <u>Delayed LB Transfer Employee</u> means an employee of VS Parent or one of its subsidiaries who transfers to employment with BBWI or one of its subsidiaries after the Spinoff Date and is classified by BBWI or the Plan Sponsor as a Delayed LB Transfer Employee for purposes of the Employee Matters Agreement. For the avoidance of doubt, a Delayed LB Transfer Employee will experience a Separation Date under the VS Plan in connection with such a transfer.

2.16. <u>Delayed VS Transfer Employee</u> means an employee of BBWI or one of its subsidiaries who transfers to employment with VS Parent or one of its subsidiaries after the Spinoff Date and is classified by BBWI or the Plan Sponsor as a Delayed VS Transfer Employee for purposes of the Employee Matters Agreement. For the avoidance of doubt, a Delayed VS Transfer Employee will experience a Separation Date under this Plan in connection with such a transfer."

5. The reference to "Section 2.16" in Section 2.16 of the Plan (renumbered as Section 2.18 by the foregoing) is hereby deleted and replaced with a reference to "Section 2.18."

6. Section 2.17 of the Plan (renumbered as Section 2.19 by the foregoing) is hereby amended by inserting a new sentence at the end thereof to read as follows:

"In the case of a Delayed LB Transfer Employee, Eligibility Computation Period will include corresponding periods credited under the VS Plan prior to his or her Delayed

Transfer Date (as defined in the Employee Matters Agreement), to the extent not already credited under this Plan."

7. Article II of the Plan is hereby amended by (i) inserting a new Section 2.20 immediately following existing Section 2.17 (renumbered as Section 2.19 by the foregoing) to read as follows and (ii) renumbering the succeeding sections of Article II accordingly:

"2.20 <u>Employee Matters Agreement</u> means that certain Employee Matters Agreement, dated as of August 2, 2021, entered into by and between L Brands, Inc. and VS Parent in connection with the Spinoff."

8. Section 2.20 of the Plan (renumbered as Section 2.23 by the foregoing) is hereby amended to read as follows:

"2.23 <u>Employer Securities</u> means stock or other securities of BBWI or its former affiliates (other than VS and its affiliates) permitted to be held by the Plan under ERISA and the Code."

9. Section 2.27 of the Plan (renumbered as Section 2.30 by the foregoing) is hereby amended by inserting a new sentence at the end thereof to read as follows:

"'Hours of Service' (as such term is defined in the VS Plan) accrued under the VS Plan prior to the applicable Delayed Transfer Date (as defined in the Employee Matters Agreement) shall be included in the computation of Hours of Service for purposes of this Plan for a Delayed LB Transfer Employee, to the extent not already credited under this Plan."

10. (i) Section 2.32 of the Plan (renumbered as Section 2.35 by the foregoing) is hereby amended to read as follows and (ii) each reference to the defined term "L Brands" in the Plan is hereby deleted and replaced with a reference to "BBWI" where the context requires:

"2.35 BBWI means Bath & Body Works, Inc., a Delaware corporation (known as L Brands, Inc. prior to August 2, 2021), and any successor thereto."

11. Section 2.39 of the Plan (renumbered as Section 2.42 by the foregoing) is hereby amended to read as follows:

"2.42 <u>Plan</u> means the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan, a profit sharing plan containing a cash or deferred arrangement, as set forth herein and as the same may from time to time be amended."

12. The references to "Section 2.63" in Section 2.62 of the Plan (renumbered as Section 2.65 by the foregoing) are hereby deleted and replaced with references to "Section 2.65."

13. The reference to "Section 2.64" in Section 2.63 of the Plan (renumbered as Section 2.66 by the foregoing) is hereby deleted and replaced with a reference to "Section 2.66."

14. Article II of the Plan is hereby amended by (i) inserting new Sections 2.67 and 2.68 immediately following existing Section 2.63 (renumbered as Section 2.66 by the foregoing) to read as follows and (ii) renumbering the succeeding sections of Article II accordingly:

"2.67. <u>Transferred Associates</u> means individuals who are employed by VS Parent or one of its subsidiaries as of the Spinoff Date (including individuals who are on a leave of absence as of the Spinoff Date).

2.68. <u>Transferred Participants</u> means Transferred Associates who are active participants in the Plan immediately prior to the Spinoff Date. For the avoidance of doubt, a Transferred Participant will not incur a Separation Date as a result of the Spinoff for purposes of the Transferred Participant's account under this Plan to be transferred to the VS Plan on the 401(k) Transfer Date."

15. Article II of the Plan is hereby amended by (i) inserting new Sections 2.73 through 2.77 immediately following existing Section 2.67 (renumbered as Section 2.72 by the foregoing) to read as follows and (ii) renumbering the succeeding sections of Article II accordingly:

"2.73. <u>VS</u> means VS Service Company, LLC, a Delaware limited liability company, and any successor thereto.

2.74. <u>VS Parent</u> means Victoria's Secret & Co., a Delaware corporation, and any successor thereto.

2.75. <u>VS Plan</u> means the Victoria's Secret & Co. 401(k) Savings and Retirement Plan.

2.76. <u>VS Securities</u> means common stock of VS Parent.

2.77. <u>VS Securities Fund</u> means the Investment Fund consisting of VS Securities and cash or cash equivalents needed to meet the obligations of such fund."

16. Section 2.68 of the Plan (renumbered as Section 2.78 by the foregoing) is hereby amended by inserting a new sentence at the end thereof to read as follows:

"A 'Year of Eligibility Service' (as such term is defined in the VS Plan) accrued under the VS Plan prior to the applicable Delayed Transfer Date (as defined in the Employee Matters Agreement) shall be included in the computation of Year of Eligibility Service for purposes of this Plan for a Delayed LB Transfer Employee, to the extent not already credited under this Plan."

17. Section 2.69 of the Plan (renumbered as Section 2.79 by the foregoing) is hereby amended by inserting a new sentence at the end thereof to read as follows:

"A 'Year of Vesting Service' (as such term is defined in the VS Plan) accrued under the VS Plan prior to the applicable Delayed Transfer Date (as defined in the Employee Matters Agreement) shall be included in the computation of Year of Vesting Service for purposes of this Plan for a Delayed LB Transfer Employee to the extent not already credited under this Plan."

18. Section 3.1(b) of the Plan is hereby amended by inserting a new sentence immediately following the first sentence thereof to read as follows:

"For the avoidance of doubt, effective as of the Spinoff Date, all Transferred Participants shall no longer be eligible to make Pre-Tax Savings Contributions, Roth Savings Contributions, Rollover Contributions and Roth Rollover Contributions to the Plan, or to receive Retirement Contributions, Matching Contributions, allocations of forfeitures or any other Employer contributions under the Plan."

19. Section 4.1(d) of the Plan is hereby amended by inserting a new paragraph at the end thereof to read as follows:

"For the avoidance of doubt, a Transferred Participant (other than as provided below for a rehired Associate) will not be eligible to receive a Retirement Contribution under this Plan for any portion of the 2021 calendar year. In the event a Transferred Participant is rehired by BBWI or its Affiliates during the 2021 calendar year after the Spinoff Date and such Transferred Participant becomes eligible for a Retirement Contribution for 2021 pursuant to this Section 4.1 upon such rehire, then for purposes of this Section 4.1 for the 2021 calendar year, such Transferred Participant's Compensation shall not include any amounts earned prior to such Participant's re-entry date provided for under Section 3.1 of the Plan."

20. The second paragraph of Section 4.1(h) of the Plan is hereby amended by inserting the following language at the end thereof to read as follows:

"For purposes of this Section 4.1(h), a Delayed VS Transfer Employee shall be deemed to experience an involuntary termination 'Without Cause' upon his or her Delayed Transfer Date (as defined in the Employee Matters Agreement). For the avoidance of doubt, a Transferred Participant will not be deemed to experience an involuntary termination 'Without Cause' under this Plan upon the Spinoff Date." 21. The third sentence of Section 4.4(a) of the Plan is hereby amended to read as follows:

"A Rollover Contribution received directly from another Eligible Retirement Plan must be paid to the Trustee in cash."

22. The third sentence of Section 4.4(b) of the Plan is hereby amended to read as follows:

"A Roth Rollover Contribution received directly from another Eligible Retirement Plan must be paid to the Trustee in cash."

- 23. The reference to "Section 2.63" in Section 4.11 of the Plan is hereby deleted and replaced with a reference to "Section 2.65."
- 24. The references to "Section 5.4(c)" in Sections 5.1, 5.2 and 5.4(b) of the Plan are hereby deleted and replaced with references to "Sections 5.4(c) and 5.5(b)."
- 25. Section 5.3 of the Plan is hereby amended to read as follows:

"5.3 Investment Funds. Two of the Investment Funds shall be the Employer Securities Contribution Fund and the VS Securities Fund. The Retirement Plan Committee shall direct the Trustee to create and maintain three or more additional Investment Funds according to investment criteria established by the Retirement Plan Committee. The Retirement Plan Committee shall have the right to direct the Trustee to merge or modify any existing Investment Funds, or to create additional Investment Funds, after notice to Participants investing in the affected Investment Funds. The Trust Fund shall be divided into (a) the Employer Securities Contribution Fund, (b) the VS Securities Fund, (c) Investment Funds established by the Retirement Plan Committee, and (d) self-directed accounts ('Self-Managed Brokerage Accounts'). Self-Managed Brokerage Account shall be treated as Investment Funds for all purposes under the Plan except that the investment authority with respect to each Self-Managed Brokerage Account shall belong to the Participant who established the Self-Managed Brokerage Account rather than the Retirement Plan Committee or any other fiduciary. No person who is otherwise a fiduciary or involved in the administration of the Plan shall have any responsibility to review any Self-Managed Brokerage Account or any investments therein. Whether a particular investment is within the category of one or another of such Investment Funds shall be determined by the Retirement Plan Committee solely in its discretion."

26. Article V of the Plan is hereby amended by (i) adding a new Section 5.5 immediately following Section 5.4 thereof to read as follows and (ii) renumbering the succeeding sections of Article V accordingly:

"5.5 Investment in VS Securities.

(a) An Investment Fund consisting primarily of VS Securities shall be maintained in the Plan. VS Securities held in the VS Securities Fund shall be held by the Trustee until distribution or sale for distribution of cash to Participants or Beneficiaries or until disposition is required to implement changes in investment designations.

(b) The VS Securities Fund shall be a 'frozen' fund initially holding VS Securities received in connection with the Spinoff. Except for amounts invested in VS Securities as of the Spinoff Date, no portion of a Participant's Account may be invested in the VS Securities Fund. A Participant may not make an investment direction that (i) any future contributions to the Participant's Account be invested in the VS Securities Fund or (ii) any portion of the existing balance in the Participant's Account be transferred to the VS Securities Fund. A Participant may, however, request a change in investment direction for the Participant's existing balance in the VS Securities Fund to be transferred to any other investment option pursuant to the rules for investment directions described in Section 5.1. Cash dividends, if any, received on VS Securities allocated to a Participant's Account shall be invested according to the Participant's investment directions in accordance with Sections 5.1 and 5.2 and shall not be reinvested in VS Securities.

(c) Notwithstanding any provision of this ARTICLE V to the contrary, but subject to Sections 5.4(c) and 5.5(b), each Participant shall have the right no less frequently than once per calendar quarter to make or change an investment direction with respect to the portion of the balance of the Participant's Pre-Tax Savings Account, Post-Tax Savings Account and Roth Savings Account that is invested in VS Securities and each Participant with at least three Years of Vesting Service shall have the right no less frequently than once per calendar quarter to make or change an investment direction with respect to the portion of the balance of the Participant's Matching Account and Retirement Account that is invested in VS Securities. In addition, all or any portion of any other Investment Funds (other than the Self-Managed Brokerage Account) may consist of VS Securities."

27 The references to "Section 5.5" in Section 5.5 of the Plan (renumbered as Section 5.6 by the foregoing) are hereby deleted and replaced with references to "Section 5.6."

28. Article V of the Plan is hereby amended by (i) adding a new Section 5.7 immediately following existing Section 5.5 (renumbered as Section 5.6 by the foregoing) to read as follows and (ii) renumbering the succeeding sections of Article V accordingly:

"5.7 <u>Voting VS Securities</u>. The Retirement Plan Committee shall have the power to direct the Trustee in the voting of all VS Securities held by the Trustee. All voting of VS Securities shall be in compliance with all applicable rules and regulations of the Securities and Exchange Commission and all applicable rules of or any agreement with

any stock exchange on which the VS Securities being voted are traded. Notwithstanding the foregoing, the Retirement Plan Committee may, in its sole discretion and at any time or from time to time, permit Participants and Beneficiaries to direct the manner in which all or the vested portion of any VS Securities allocated to their Accounts shall be voted on such matters as the Retirement Plan Committee permits. Upon timely receipt of directions under this Section 5.7 from the Retirement Plan Committee, Participant or Beneficiary, the Trustee shall vote all VS Securities as directed. If the Trustee does not receive timely directions from the Retirement Plan Committee, Participant or Beneficiary under this Section 5.7, the Trustee shall not vote the VS Securities with respect to which direction was not given."

29. Section 5.6 of the Plan (renumbered as Section 5.8 by the foregoing) is hereby amended to read as follows:

"5.8 <u>Tender Offers</u>. Each Participant and Beneficiary shall have the sole right to direct the Trustee as to the manner in which to respond to a tender or exchange offer for Employer Securities or VS Securities, as applicable, allocated to such person's Account. The Retirement Plan Committee shall use its best efforts to notify or cause to be notified each Participant and Beneficiary of any tender or exchange offer and to distribute or cause to be distributed to each Participant and Beneficiary such information as is distributed in connection with any tender or exchange offer to holders generally of Employer Securities or VS Securities, as applicable, together with the appropriate forms for directing the Trustee as to the manner in which to respond to such tender or exchange offer. Upon timely receipt of directions under this Section 5.8 from the Participant or Beneficiary, the Trustee shall respond to the tender or exchange offer in accordance with, and only in accordance with, such directions. If the Trustee does not receive timely directions from a Participant or Beneficiary under this Section 5.8, the Trustee shall not tender, sell, convey or transfer any Employer Securities or VS Securities, as applicable, allocated to such person's Account in response to any tender or exchange offer."

30. The reference to "Section 5.7" in Section 5.7 of the Plan (renumbered as Section 5.9 by the foregoing) is hereby deleted and replaced with a reference to "Section 5.9."

31. Section 7.1(d) of the Plan is hereby amended to read as follows:

"(d) Notwithstanding anything in this Section 7.1 to the contrary, (i) if a Participant dies on or after January 1, 2007 while performing qualified military service (as defined in Code Section 414(u)), the Participant's Retirement Account shall become fully vested and nonforfeitable upon the date of his death, provided that the Participant would have been eligible for reinstatement of employment with the Employer had the Participant's qualified military service ended on the day before his death, (ii) if a

8

Participant's employment is transferred from an Employer to TSAM (Delaware) LLC, prior to October 31, 2011, in connection with the initial formation of TSAM (Delaware) LLC, the Participants' Retirement Account shall become fully vested and nonforfeitable on the date of such transfer; and (iii) a Delayed VS Transfer Employee's Retirement Account shall become fully vested and nonforfeitable upon his or her Delayed Transfer Date (as defined in the Employee Matters Agreement)."

32. Section 8.1(a) of the Plan is hereby amended to read as follows:

"(a) Except as otherwise elected in accordance with Section 8.2(h), a Participant or Beneficiary shall receive any benefits to which he or she is entitled in the form of a single-sum distribution consisting of cash, Employer Securities, and/or VS Securities. Amounts not invested in either Employer Securities or VS Securities shall be distributed in cash. Amounts invested in the Employer Securities Contribution Fund shall be distributed in cash, unless the Participant or Beneficiary elects to receive the Employer Securities Contribution Fund in whole shares of the Employer Securities (plus cash for any fractional shares). Amounts invested in the VS Securities (plus cash for any fractional shares) fund in whole shares of the VS Securities (plus cash for any fractional shares)."

33. Section 8.1(d) of the Plan is hereby amended by deleting the phrase "Employer Securities" and replacing it with the phrase "Employer Securities or VS Securities" wherever it appears therein.

34. Section 8.5(a) of the Plan is hereby amended by deleting the phrase "Employer Securities" and replacing it with the phrase "Employer Securities or VS Securities, as applicable," wherever it appears therein.

35. Section 8.5(d) of the Plan is hereby amended to read as follows:

"(d) A request for an in-service withdrawal shall be made in accordance with such procedures and on such forms as the Retirement Plan Committee may determine. An in-service withdrawal shall be paid in cash and/or whole shares of Employer Securities or VS Securities, as applicable, as elected by the Participant, as soon as practicable after the Participant requests an in-service withdrawal. The amount of any in-service withdrawal made to a Participant, whether paid in cash, Employer Securities, VS Securities, or a combination thereof, shall not be less than \$500."

36. Section 8.5 of the Plan is hereby amended by inserting a new subsection (g) at the end thereof to read as follows:

"(g) Notwithstanding any provision of the Plan to the contrary and solely for purposes of determining eligibility to receive in-service withdrawals under this

9

Section 8.5, a Transferred Participant will be treated as an active Participant in this Plan for purposes of in-service withdrawals made pursuant to this Section 8.5 during the period beginning on the Spinoff Date and ending on the 401(k) Transfer Date."

IN WITNESS WHEREOF, the Company has caused this First Amendment to be executed by its duly authorized officer this 28th day of July, 2021.

L BRANDS, INC.

By: /s/ Deon Riley

Deon Riley Chief Human Resources Officer

SECOND AMENDMENT TO THE BATH & BODY WORKS, INC. 401(k) SAVINGS AND RETIREMENT PLAN (AS AMENDED AND RESTATED EFFECTIVE APRIL 5, 2021)

WHEREAS, L Brands Service Company, LLC, formerly known as Limited Brands Service Company, LLC (the "Company") sponsors the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan, as amended and restated effective April 5, 2021 (the "Plan"), for the benefit of eligible associates of the Company and its affiliates; and

WHEREAS, the Company desires to amend the Plan to make certain changes to the definition of "Compensation";

NOW, THEREFORE, BE IT RESOLVED, effective as of January 1, 2022, the Plan is hereby amended as follows:

1. The first paragraph of Section 2.13 of the Plan is hereby amended to add a new sentence at the end thereof to read as follows:

"Notwithstanding the foregoing, (a) bonuses (other than sign-on bonuses or retention bonuses) paid after a Participant's severance from employment shall be included as Compensation for purposes of the calculation of Retirement Contributions under Section 4.1 and not for purposes of any other contribution under the Plan, and (b) effective as soon as administratively practicable after a Participant's severance from employment, Compensation shall not include any other payments made after a Participant's severance from employment."

IN WITNESS WHEREOF, the Company has caused this Second Amendment to be executed by its duly authorized officer this 29th day of December, 2021.

L BRANDS, INC.

By: /s/ Deon Riley Deon Riley

Chief Human Resources Officer

Davis Polk

Davis Polk & Wardwell LLP 450 Lexington Avenue New York, NY 10017 davispolk.com

February 10, 2022

EXHIBITS 5.1 AND 23.2

OPINION OF DAVIS POLK & WARDWELL LLP

Bath & Body Works, Inc. Three Limited Parkway Columbus, Ohio 43230

Ladies and Gentlemen:

We have acted as special counsel to Bath & Body Works, Inc., a Delaware corporation (the "**Company**"), and are delivering this opinion in connection with the Company's Registration Statement on Form S-8 (the "**Registration Statement**") filed with the Securities and Exchange Commission pursuant to the Securities Act of 1933, as amended, for the registration of 500,000 shares of the Company's common stock, par value \$0.50 per share (the "**Shares**"), issuable pursuant to the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan (f/k/a the L Brands, Inc. 401(k) Savings and Retirement Plan) (the "**Plan**").

We, as your counsel, have examined originals or copies of such documents, corporate records, certificates of public officials and other instruments as we have deemed necessary or advisable for the purpose of rendering this opinion.

In rendering the opinion expressed herein, we have, without independent inquiry or investigation, assumed that (i) all documents submitted to us as originals are authentic and complete, (ii) all documents submitted to us as copies conform to authentic, complete originals, (iii) all signatures on all documents that we reviewed are genuine, (iv) all natural persons executing documents had and have the legal capacity to do so, (v) all statements in certificates of public officials and officers of the Company that we reviewed were and are accurate and (vi) all representations made by the Company as to matters of fact in the documents that we reviewed were and are accurate.

On the basis of the foregoing, we are of the opinion that the Shares have been duly authorized and, when and to the extent issued pursuant to the Plan upon receipt by the Company of the consideration for the Shares specified therein, if and as applicable, will be validly issued, fully paid and non-assessable.

We are members of the Bar of the State of New York and the foregoing opinion is limited to the General Corporation Law of the State of Delaware.

Davis Polk

We hereby consent to the filing of this opinion as an exhibit to the Registration Statement. In giving this consent, we do not admit that we are in the category of persons whose consent is required under Section 7 of the Securities Act.

Very truly yours,

/s/ Davis Polk & Wardwell LLP

DEPARTMENT OF THE TREASURY

INTERNAL REVENUE SERVICE P. 0. BOX 2508 CINCINNATI, OH 45201



THREE LIMITED PKY

COLUMBUS, OH 43230

Employer Identification Number: 31-1048997 DLN: 17007306054022 LIMITED BRANDS SERVICE COMPANY LLC Person to Contact: THERESE M BOWDREN ID# 31213 Contact Telephone Number: (513) 263-3895 Plan Name: LIMITED BRANDS INC 401K SAVINGS AND RETIREMENT PLAN Plan Number: 002

Dear Applicant:

We have made a favorable determination on the plan identified above based on the information you have supplied. Please keep this letter, the application forms submitted to request this letter and all correspondence with the Internal Revenue Service regarding your application for a determination letter in your permanent records. You must retain this information to preserve your reliance on this letter.

Continued qualification of the plan under its present form will depend on its effect in operation. See section 1.401-1(b)(3) of the Income Tax Regulations. We will review the status of the planin operation periodically.

The enclosed Publication 794 explains the significance and the scope of this favorable determination letter based on the determination requests selected on your application forms. Publication 794 describes the information that must be retained to have reliance on this favorable determination letter. The publication also provides examples of the effect of a plan's operation on its qualified status and discusses the reporting requirements for qualified plans. Please read Publication 794.

This letter relates only to the status of your plan under the Internal Revenue Code. It is not a determination regarding the effect of other federal or local statutes.

This determination letter gives no reliance for any qualification change that becomes effective, any guidance published, or any statutes enacted, after the issuance of the Cumulative List (unless the item has been identified in the CUmulative List) for the cycle under which this ication was submitted.

This determination letter is for the amendment(s) executed on 8/7/12 &10/27/11.

This determination letter is also applicable for the amendment(s) dated on 12/30/10 &12/14/10.

This determination letter is also

for the amendment(s) dated on

Letter 2002 (DO/CG)

- 1 T I

LIMITED BRANDS SERVICE COMPANY LLC

1/13/10 &6/24/09.

This determination is subject to your adoption of the proposed amendments submitted in your letter dated 8/29/13. The proposed amendments should be adopted on or before the date prescribed by the regulations under Code section 401(b).

This letter may not be relied on after the end of the plan's first five-year remedial amendment cycle that ends more than 12 months after the application was received. This letter expires on January 31, 2018. This letter considered the 2011 Cumulative List of Changes in Plan Qualification Requirements.

The information on the enclosed addendum is an integral part of this determination. Please be sure to read and keep it with this letter.

We have sent a copy of this letter to your representative as indicated in the Form 2848 Power of Attorney or appointee as indicated by the Form 8821 Tax Information Authorization.

If you have questions concerning this matter, please contact the person whose name and telephone number are shown above.

Andrew E. Zuckerman Director, EP Rulings &Agreements

Enclosures: Publication 794 Addendum

Letter 2002 (DO/CG)

LIMITED BRANDS SERVICE COMPANY LLC

This determination letter is also applicable to the amendments dated $3/2/09,\ 11/5/08,\ \text{and}\ 11/8$

. т. *з*у Т

To the Board of Directors and Shareholders of Bath & Body Works, Inc.

We are aware of the incorporation by reference in the Registration Statement (Form S-8) of Bath & Body Works, Inc. for the registration of 500,000 shares of its common stock of our reports dated June 3, 2021, September 2, 2021, and December 3, 2021 relating to the unaudited condensed consolidated interim financial statements of Bath & Body Works, Inc. (formerly known as L Brands, Inc.) that are included in its Forms 10-Q for the quarters ended May 1, 2021, July 31, 2021 and October 30, 2021.

/s/ Ernst & Young LLP

Grandview Heights, Ohio February 10, 2022

Consent of Independent Registered Public Accounting Firm

We consent to the incorporation by reference in the Registration Statement (Form S-8) pertaining to the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan of our reports dated March 19, 2021, with respect to the consolidated financial statements of Bath & Body Works, Inc. (formerly known as L Brands, Inc.), and the effectiveness of internal control over financial reporting of L Brands, Inc., included in its Annual Report (Form 10-K) for the year ended January 30, 2021, and dated June 29, 2021, with respect to the financial statements and schedule of the Bath & Body Works 401(k) Savings and Retirement Plan (formerly known as the L Brands, Inc. 401(k) Savings and Retirement Plan) included in its Annual Report (Form 11-K) for the year ended December 31, 2020, filed with the Securities and Exchange Commission.

/s/ Ernst & Young LLP

Grandview Heights, Ohio February 10, 2022

Calculation of Filing Fee Tables

FORM S-8 Registration Statement Under the Securities Act of 1933

(Form Type)

Bath & Body Works, Inc.

(Exact Name of Registrant as Specified in its Charter)

Table 1: Newly Registered Securities							
Security Type	Security Class Title (1)	Fee Calculation Rule	Amount Registered (1)	Proposed Maximum Offering Price Per Unit (2)	Maximum Aggregate Offering Price (1)(2)	Fee Rate	Amount of Registration Fee (3)
Equity	Common Stock, \$0.50 par value per share (1)	457(c) and 457(h)	500,000	\$55.77	\$27,885,000	0.0000927	\$2,584.94
Total Offering Amounts					\$27,885,000		\$2,584.94
Total Fee Offsets							-
Net Fee Due							\$2,584.94

(1) Pursuant to Rule 416(a) under the Securities Act of 1933, as amended (the "Securities Act"), this Registration Statement on Form S-8 (this "Registration Statement") also covers additional shares of common stock, \$0.50 par value per share ("Common Stock"), of Bath & Body Works, Inc. (the "Registrant") that may be offered or issued under the Bath & Body Works, Inc. 401(k) Savings and Retirement Plan (f/k/a the L Brands, Inc. 401(k) Savings and Retirement Plan) (the "Plan") as a result of any stock dividend, stock split or other similar transaction. In addition, pursuant to Rule 416(c) under the Securities Act, this Registration Statement also covers an indeterminate amount of interests to be offered or sold pursuant to the Plan.

(2) Estimated solely for purposes of calculating the registration fee pursuant to Rule 457(c) and Rule 457(h) under the Securities Act based on the average of the high and low prices of Common Stock as reported on the New York Stock Exchange on February 4, 2022. Pursuant to Rule 457(h)(2) under the Securities Act, no separate fee is required to register plan interests pursuant to this Registration Statement.

(3) Rounded up to the nearest penny.